# THE EFFECTS OF SCALING AND ORAL HYGIENE EDUCATION ON THE PERIODONTAL STATUS OF FACTORY WORKERS IN KENYA

BY

DR. NYOKABI MARGARET MACHARIA

B.D.S., NAIROBI

A DISSERTATION SUBMITTED TO:

DEPARTMENT OF COMMUNITY HEALTH COLLEGE OF HEALTH SCIENCES UNIVERSITY OF NAIROBI IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF MASTERS OF PUBLIC HEALTH.

UNIVERSITY OF NAIROBI

MEDICAL LIBRARY UNIVERSITY OF NAIROBI



#### DECLARATION

I, Nyokabi Margaret Macharia, do hereby state that this dissertation is my original work and has never been submitted before, either in part or in whole, for another degree in any other University.

This dissertation has been submitted to the University of Nairobi for examination with our approval as supervisors.

Signed ........

FIROZE MANJI, B.D.S., M.Sc., Ph.D.

Regional representative for Eastern and Souther Africa.

un .

Health Science Division

International Development Research Centre

LAly about a

Signed ..

LAMBERT NYABOLA, B.Sc., M.Sc., M.S., Dip. Epid.

Lecturer

Department of Community Health

University of Nairobi

> JOSEPH K. WANG'OMBE., B.A., M.A., Ph.D. Senior Lecturer

Department of Community Health

University of Nairobi.

Signed

iii

Elisha.K. Muchunga.,B.A.,M.P.H.,Ph.D. Chairman

Department of Community Health

University of Nairobi

This dissertation is dedicated to my parents.

#### ACKNOWLEDGEMENT

I wish to express my most sincere gratitude to all those who in one way or another assisted in this study.

I would like to especially thank the following:

- The management of Unga Ltd and Unga Maize Millers Ltd, for allowing me to carry out this research in their factories. I would also like to extend my thanks to the nurses and factory workers for their participation and support.
- The Director of Kenya Medical Research Institute (KEMRI) for allowing me to work on attachment at the institute for the duration of the study. This gave me access to facilities that greatly assisted in the analysis of data in this study.
  The International Development Research Centre (IDRC) and Colgate-Palmolive (EA) Ltd. for their financial assistance.

 Last but not the least, my supervisors for their guidance and patience.

V

# TABLE OF CONTENTS

Title	Page
Declaration	ii
Dedication	iv
Acknowledgement	v
Table of contents	vi
List of Tables	xi
List of Appendices	xvi
Abstract	xvii
Introduction to chapters	xxi

# CHAPTER ONE

INTRO	DUCTION1
1	Background1
1.2	Statement of problem and justification
	of the study
1.3	Aims and objectives8
1.4	Hypothesis

# CHAPTER TWO

2.	LITERATURE REVIEW	10
2.1	Oral hygiene status and	
	periodontal disease	10
2.2	Calculus and periodontal disease	11
2.3	Pathogenesis of periodontal disease	13

2.4	Effect of oral hygiene measures	
	on periodontal status	14
2.5	Effect of scaling on periodontal	
	status	16

# CHAPTER THREE

3.	METHODS AND MATERIALS	19
3.1	Study area	19
3.2	Study design	20
3.3	Study population	21
3.4	Sample size determination and	
	randomization	21
3.5	Sample selection and randomization	22
3.6	Eligibility criteria	22
	3.6.1. Inclusion criteria	22
	3.6.2. Exclusion criteria	23
3.7	Ethical considerations	23
3.8	Data collection	23
3.9	Oral examination	24
3.10	Measurement Criteria	26
3.11	Treatments	28
	3.11.1 Scaling	28
	3.11.2 Oral hygiene education	29
3.12	Schedule of the study	29
3.13	Intra-observer reliability	30

vii

37	7	٦.	٦.
Ý	1	1	1

3.14 Ana	alysis.						31
----------	---------	--	--	--	--	--	----

# CHAPTER FOUR

4.	RESULTS		.32
4.1	General c	haracteristics of the study	
	populatio	m	32
4.2	Distribut	ion of study subjects in the treatment	
	groups		34
4.3	Compariso	on of the groups at the	
	baseline	examination	35
	4.3.1	Plaque	35
	4.3.2	Calculus	37
	4.3.3	Gingivitis	39
	4.3.4	Pocket depth	41
	4.3.5	Attachment level	42
4.4	The diffe	rences that occurred	
	within t	he groups between the	
	baseline	and final examinations	. 44
	4.4.1	Plaque	. 44
	4.4.2	Calculus	. 45
	4.4.3	Gingivitis	47
	4.4.4	Pocket depth	48
	4 4 5	Attachment level	. 49

.5	Between g	roups comparison of the	
	differenc	es that occurred within them	
	after tre	atment	
	4.5.1	Plaque	ł
	4.5.2	calculus	
	4.5.3	Gingivitis 59	1
	4.5.4	Pocket depth 63	
	4.5.5	Attachment level	

# CHAPTER FIVE

э.	DISCUSSIO	N, CONCLUSION, RECOMMENDATIONS	
	AND LIMIT	ATIONS OF STUDY72	2
5.1	DISCUSSIO	N	
	5.1.1	General characteristics72	
	5.1.2	Distribution of participants in	
		the treatment groups73	
	5.1.3	Comparison of the groups at	
		baseline	
	5.1.4	Differences in periodontal	
		parameters within the four groups	
		between the baseline and final	
		examinations	
	5.1.5	Effects of scaling and OHE	
		on periodontal parameters	
5.2	Conclusio	ns	
5.3	Recommend	lations	

ix

5.4	Limit	ati	ons	Ō	f	t	he		S	tι	ıd	y		•	•			 •		,	,	•	•			٠	86
Refer	rences			. ,	. ,			,	•						•	 ,	•	 •		,	,					•	88
Apper	ndices				• •			,	•			•	• •			 •		 ٠	• •		٠	•	•	, ,	٢	•	95

Х

# LIST OF TABLES

1.	Frequency distribution of subjects
	in the treatment groups
2.	Percentage distribution of
	subjects by treatment groups
3.	a) Mean number of sites with plaque
	score 0,1,2,3 and standard deviations
	at baseline by treatment groups
	b) F and p values for the differences in plaque scores
	between treatment groups at
	baseline
4.	a) The mean number of sites with calculus
	score 0,1,2,3 at baseline by
	treatment groups
	b) The F and p values for the differences
	in calculus between treatment groups
	at baseline
5.	a) Mean number of sites with gingivitis
	score 0,1,2,3 in the treatment groups at
	baseline
	b) The F and p values for the differences
	in gingivitis between treatment groups at
	baseline40
6.	a) Mean number of sites with shallow
	( $\leq 3$ mm) and deep ( $\geq 4$ mm) pockets at
	baseline

xi

	b) F and p values for the differences
	in pocket depth between the treatment groups
	at baseline41
7.	a) Mean number of sites with attachment
	1evel 0 mm,1-3 mm, ≥4 mm42
	b) F and p values for differences
	in attachment level between the
	groups at baseline43
8.	Means of the differences in the
	number of sites with plaque score 0,1,2,3
	in the treatment groups45
9.	Means of differences in the
	number of sites with calculus score
	0,1,2,3 in the treatment
	groups
10.	Means of differences in the
	number of sites with gingivitis score
	0,1,2,3 in the treatment
	groups
11.	Means of differences in the
	number of sites with shallow ( $\leq$ 3 mm) and
	deep (2 4 mm) pockets in the treatment groups48
12.	Means of differences in the
	number of sites attachment level 0 mm,
	1-3 mm, and > 4 mm in the treatment groups

xii

13.	Means of differences in the
	plaque score 0,1,2,3 in the "scaled"
	and "non scaled" groups
14.	Means of differences in plaque
	score 0,1,2,3 in the "OHE" and
	"no OHE"groups
15.	Means of differences plaque score 0,1,2,3
	in "scaled groups" (1 and 2) and
	the "non-scaled" groups (3 and 4)53
16.	Means of differences in sites with
	plaque score 0,1,2,3 in groups with
	OHE (2 and 3) and without
	OHE (1 and 4)54
17.	Means of differences in number of
	sites with calculus score
	0,1,2,3 in the "scaled' (1 and 2)
	and "non scaled" (3 and 4) groups
18.	Means of the differences in the number of
	sites with calculus score
	0,1,2,3 in the groups with OHE (2 and 3) and
	without OHE (1 and 4)
19.	Means of differences in the
	number of sites with calculus score
	0,1,2,3, in scaled group (1 and 2) and
	non scaled (3 and 4) groups

xiii

20. Means of differences in number of sites with calculus score 0,1,2,3 in groups with OHE (2 and 3) and without OHE (1 and 4).....59 21. Means of differences in number of sites with gingivitis score 0,1,2,3, in the scaled (1 and 2) and non scaled 22. Means of differences in number of sites with gingivitis score 0, 1, 2, 3 in groups with (2 and 3)and without OHE (1 and 4).....61 23. A comparison of means of differences in number of sites with gingivitis score 0,1,2,3 between groups 1, 2 (scaled) and 3,4 (non scaled).....62 24. Means of differences in number of sites with gingivitis score 0,1,2,3 in OHE (3), scaling combined with OHE (2) and the no OHE (1 and 4) groups..... 63 25. Means of differences in number of sites with shallow (≤3 mm) and deep (24 mm) pockets in the scaled (1 and 2) and 26. Means of the differences in number of sites with shallow (≤3 mm) and

xiv

deep (≥4 mm) pockets in the "OHE" and Means of differences in number of 27. sites with shallow ( $\leq 3$  mm) and deep ( $\geq 4$  mm) pockets in "scaled" (1,2) and "non scaled" 28, Means of differences in number of sites with shallow (≤3 mm) and deep (24 mm) pockets in groups with (2,3)and without OHE (1,4).....67 Means of differences in number 29. of sites with attachment level 0 mm, 1-3 mm and ≥4 mm in the "scaled" and "non scaled" Means of differences in number of 30. sites with attachment level 0 mm, 1-3 mm and ≥4 mm in the "OHE" and Means of differences in number of 31. sites with attachment level 0 mm, 1-3 mm and ≥4 mm in the "scaled" Means of the differences in number 32. of sites with attachment level 0 mm, 1-3 mm and ≥4 mm in groups with 2,3 and the "no OHE"

χv

# xvi

# LIST OF APPENDICES

Ι.	History sheet95
II.	Data collection form
III.	Guidelines for oral hygiene
	instructions99
IV.	Intra-observer reliability101

#### ABSTRACT

This study was designed to determine the effects of scaling and of oral hygiene education (OHE) on the periodontal status of a group of 101 factory workers.

101 male factory workers aged between 30-50 years were selected from 2 milling factories in the Nairobi's industrial area. These workers were first subjected to a baseline examination after which they were randomly assigned to 4 intervention groups. The treatments were carried out three months after the baseline examination. Group 1, received scaling alone, group 2, scaling combined with OHE, group 3 OHE alone and group 4 received no treatment at all. A final examination was carried out 6 months following treatment.

There were no differences among the groups at baseline. For the final examination, there were statistically significant differences among the groups in: plaque score 2(F=4.24; P=0.008), gingivitis score 0(F=3.31; P=0.0246), supragingival calculus (F=3.68; P=0.0158), subgingival calculus (F=41.48; P=0.0001) and pocket depth (F=2.87; P=0.042).

An analysis of the changes that occurred within each group between the baseline and final examination indicated that there was a significant increase in the number of plaque free sites in all the treatment groups, p<0.001, p<0.01, p<0.01, and p<0.05 for

xvii

groups 1,2,3, and 4 respectively. The number of calculus free sites increased significantly, p<0.001 in both groups 1 and 2. This was mainly due to a decrease in the number of sites with subgingival calculus. These decreased significantly (p<0.001) in these groups, while in the groups that did not receive scaling i.e groups 3 and 4 the number of sites with subgingival calculus increased significantly (p<0.01 and 0.05 respectively).

The number of sites with supragingival calculus decreased significantly (p<0.01) only in group 1. A significant increase in the number of gingivitis free sites was observed in group 1 (p<0.001) and group 2 (p<0.01). Groups 1 and 2 demonstrated a significant increase in the number of shallow pockets (p<0.01, p<0.01 respectively) and a decrease in the number of deep pockets (p<0.01, p<0.05 respectively). In all the groups, the number of sites with attachment loss 0 mm increased significantly (p<0.001 in all the groups) while the number of sites with attachment loss 1-3 mm decreased significantly (p<0.001 for all the groups). However, the number of sites with attachment loss 24 mm showed a significant decrease (p<0.05) in group 2 only.

The effect of scaling was determined by comparing the "scaled" group (group 1+2) with the "non scaled" group (group 3+4). For plaque, scaling was found to have a significant effect (p<0.05) on plaque score 2 only. It had no effect on the number of plaque free sites (score 0). Scaling was found to significantly

#### xviii

(p<0.0001) increase the number of calculus free sites. This was found mainly attributed to a decrease in the number of sites with subgingival calculus (p<0.0001). Scaling did not have a significant effect on the supragingival calculus. Scaling was associated with a significant increase (p<0.003) in the number of gingivitis free sites. Scaling also caused a significant reduction (p=0.05) in the number of deep pockets. Following scaling, the number of sites with attachment loss 0 mm increased significantly (p=0.01) while the number of sites with attachment loss 1-3 mm and  $\ge 4$  mm decreased significantly (p=0.035) and p=0.021 respectively).

In this study, scaling was found to improve all the periodontal parameters except for plaque. Scaling resulted in a limited improvement on the pocket depth and attachment level. The greatest effect of scaling was limited to the subgingival calculus. Scaling produced a significant reduction in the number of bleeding sites. OHE was not effective in this study. A combination of scaling and OHE did not offer any significant advantage over scaling alone.

Although scaling was associated with a statistically significant improvement in periodontal status over a six month period study, some deterioration was observed to have occurred after treatment on the supragingival calculus. Due to the poor plague controlled in the scaled group, it is possible that a deterioration may have

xix

occurred in the other periodontal parameters. This would not have been detectable in this study since it had only one posttreatment examination. Before scaling can be recommended as a method of treating periodontal disease in the community, better methods of controlling plaque should be sought. Also, a longer study with more post-treatment examinations should be carried out to determine the effects of scaling on the progression of the disease.

 $\mathbf{X}\mathbf{X}$ 

#### INTRODUCTION TO CHAPTERS

This thesis is organised into five chapters. The first part of chapter 1 comprises of literature review on periodontal disease, the global epidemiclogy, the prevalence of the disease in Kenya, the management and a summary of the findings on pathogenesis and treatment of periodontal disease. The second part of the chapter contains a statement of the problem in the current management of periodontal disease in Kenya and justification for further studies in this field. The aims and objectives and the hypotheses of the current study constitute the last part of chapter 1. Chapter 2 contains the literature review. In this chapter, the literature has been divided into five sections. These sections include literature on the association of plaque and calculus and periodontal disease, calculus and periodontal disease, theories on the pathogenesis of periodontal disease, the effects of OHE on periodontal disease and the effect of scaling on periodontal disease.

Chapter 3 contains the methods and materials section. This chapter contains descriptions of the study area, the study design, the study population, the sample size determination, the selection and randomisation procedures. The methods and instruments used in the collection of data and the method and instruments used in the examinations indicated. The parameters and the criteria that were used to measure each parameter are

xxi

treatments. A time schedule for the study has been given. Also included in this chapter are the results of the intra-observer reliability and an outline of the analysis in this study.

Chapter 4 contains the results which are divided into five major sections. The first section includes results of the general characteristics of the population. Here the mean age and the general tooth status of the population has been given. Section 2 indicates the distribution of the number of individuals in the four treatment groups. Section 3 consists of the results of the comparison of the scores of various parameters in the four groups at the baseline examination. The fourth section consists of results of the comparison of the differences that occurred within the treatment groups between the baseline and final examination. The fifth section consists of the results of the comparison, between the "scaled" and "non scaled", between the "OHE" and "non OHE" and between the scaling only and scaling combined with OHE groups, of the differences that occurred in the various parameters after treatment.

Chapter 5 contains the discussion, conclusion, recommendations and limitations of the study. The final pages include the bibliography and the appendices.

xxii

#### CHAPTER 1

#### INTRODUCTION

This chapter contains a background on periodontal disease, statement of the problem in Kenya, aims, objectives and hypotheses.

#### 1.1 Background

Periodontal disease is a term which refers to a group of diseases that cause destruction of the periodontium. Periodontal disease has been described as the most widespread disease of mankind (WHO, 1978). Different populations have been found to have different prevalence and severity of the disease. Information from the WHO Global Oral Data Bank from surveys carried out in different countries indicate that the prevalence and severity of periodontal disease may be greater in the developing than in the developed countries. The trends indicate that in developing countries the prevalence and severity remain high unlike in the developed countries where they appear to be decreasing (WHO 1984).

Very few studies have been published on prevalence and severity of periodontal disease in Africa. However, existing literature both to be fairly high in the region (Sheiham, 1981, Baelum, 1986, 1988, Budal et al 1985, Olssen, 1978). Akapbio (1970) reported that periodontal disease is universal in the

adult African. In their study on a rural population in Kenya, Baelum et al (1988) found that the majority of adult population exhibited gross accumulations of plaque and calculus. The population studied was considered representative of other rural populations in Kenya. Therefore, considering that approximately 80% of the Kenyan population is rural, it can be assumed that periodontal disease is widespread in the adult population in this country.

Periodontal disease is characterised by poor oral hygiene with accumulation of plaque and calculus on teeth surfaces, pocket formation, loss of attachment, gingival recession and gingival bleeding. The disease leads to loss of tooth support thus resulting in increased tooth mobility and migration that ultimately lead to tooth loss. According to a WHO report (WHO 1978), it was stated that periodontal disease "deprives many of their teeth long before old age." However, studies carried out in developed and developing countries over the last ten years indicate that caries rather than periodontal disease is the major cause of tooth loss (Ainamo, 1984, Baillit 1987, Kaimenyi 1986).

Conventional periodontal therapy involves removal of plaque and calculus deposits through scaling. This has for a long time been considered to be effective in controlling periodontal disease. The impression is based on observations from epidemiological studies which suggest existence of an association

between plaque/calculus and severity of periodontal disease (Loe.H, et al,1978).

Studies in literature indicate that scaling is effective controlling periodontal disease (Ramjford,S.P et al, 1973; Lindhe.J et al, 1982). However, unless done frequently, scalin has been found not to be effective in preventing progression of periodontal disease (Axelsson 1978, Soumi 1969). These studies found that conventional periodontal therapy, where scaling is done once or twice a year, is not effective in preventing progression of disease. A few studies have however, found that scaling may cause deterioration of shallow pockets (Badersten 1981, Lindhe 1982).

The rationale for periodontal treatment has been that untreated periodontal disease progresses continuously in a lin fashion throughout the mouth and leads to an extensive amount tooth loss (WHO 1978) in all individuals. However, recent studies now indicate that the untreated disease progresses wit periods of activity and periods of remission (Socransky 1984, Goodson 1987). Other studies have shown that in a population, only a small portion of individuals are susceptible to rapidly progressive disease (Manji et al 1988, Jackson 1986). In most individuals, periodontal disease progresses slowly. This has been found true even in populations with high prevalence of po

oral hygiene conditions (Baelum et al, 1986, 1988, Loe et al, 1988).

In commenting on data collected from surveys done in different countries by the WHO, Barmes (1986) observed that different populations have different susceptibilities to periodontal disease. Baelum et al (1986) compared the results of studies on periodontal disease in different countries which also supported this view.

Studies carried out in East Africa seem to indicate that these populations are not very susceptible to the progression of periodontal disease (Manji 1988, Baelum 1986, Olsson 1978). In these populations, the rate of progression of untreated periodontal disease appears to be compatible with the retention of teeth in old age. According to Baelum (1988), this 'resistance' to periodontal disease progression may be due to the fact that the plaque and calculus deposits in these populations are undisturbed. It was stated in this study that "microbial deposits and calculus if left undisturbed may not necessarily be associated with the development of deep pathological pockets and extensive loss of attachment." If this is true then we should reexamine our management of periodontal disease in this country. Currently periodontal disease is treated by scaling. The finding that different populations may have different susceptibilities to periodontal disease is an indication that the disease may require

different ways of management in the various populations.

The fact that studies have indicated that scaling has to be done very frequently to be effective suggests that scaling is not the method of choice in treating periodontal disease on a public health basis, since it is not possible to carry out scaling more than once a year (and possibly longer) on a large population. However, before scaling can be dismissed altogether, it would be interesting to find out the effect on the periodontal status of scaling once a year as compared to giving OHE and no treatment.

Most studies on the effects of scaling on periodontal disease have been done in industrialised countries. No such studies have been published on Kenyans.

## 1.2 Statement of Problems and Justification for the Study

Although there are very few studies on prevalence of periodontal disease in Kenya, available evidence suggests that the disease is highly prevalent in the adult population (Akapbio 1970, Baelum 1988, Owino 1984). Currently, the management of periodontal disease in Kenya is by scaling. However, there have been no studies published so far on effectiveness of scaling in a *population of Kenyans. Since susceptibility to periodontal* disease may vary from one population to another, scaling might not necessarily be the best way of controlling periodontal

disease in all settings.

Published literature indicates that scaling is not effective in preventing the progression of periodontal disease unless it is performed frequently. The conventional approach to periodontal therapy ie. scaling once or twice a year, has been found to be ineffective in controlling progression of periodontal disease (Axelsson, 1978, Soumi 1969). Besides, scaling more than once a year is not practical on a public health basis; especially in a developing country where resources are scarce. The question that therefore arises is whether conventional therapy is better than no treatment at all, and whether cheaper options such as OHE can be more cost-effective.

Manji and Sheiham (1986) in a study on 9124 Kenyan children calculated that it would require 200 Kenyan dentists roughly 7-21 years of work (without follow-up) to treat just one cohort of 5-15 year olds. Considering that the adult population has a lot more deposits of calculus, it would be expected that more resources would be required to treat adults.

It is therefore important to determine the effectiveness of scaling before investing more resources in scaling as a way of controlling periodontal disease in Kenya. We should compare its effect with that of other cheaper options like oral hygiene education. Available evidence suggests that untreated periodontal disease does not always lead to extensive tooth loss (Baelum

1986, 1988, Olsson 1978) and that in some populations the rate of progression of the disease may be compatible with the retention of a functional dentition in old age in the majority of individuals (Manji 1988, Pilot 1986). Further, scaling may not always be necessary in some populations and that the improvement of oral hygiene measures through oral health education may prove to be a more cost-effective way of managing the disease. Studies in Kenya and Tanzania suggest that these populations may be "resistant" to periodontal disease and that the rate of progression of periodontal disease is relatively slow.

In the present study, the effects of scaling and oral hygiene education on the periodontal disease status of a group of Kenyan factory workers was determined and compared.

#### 1.3 Aims and Objectives

#### Aims of the Study

The aim of this study is to determine the effect of scaling and oral health education (OHE) on periodontal status on an adult Kenyan population with minimal access to dental care over a 6 months follow-up period.

#### OBJECTIVES

- To determine the effect of scaling alone on the amount of plaque, calculus, pocket depth, attachment loss and gingival bleeding.
- To determine the effect of oral hygiene education alone on the amount of plaque, calculus, pocket depth, attachment loss and gingival bleeding.
- To determine the effect of combining scaling and oral hygiene education on the above indicators of periodontal status.

## 1.4 HYPOTHESES (Alternate).

- Scaling will produce a significant reduction in all the periodontal parameters.
- OHE will produce a significant reduction in the amount of plaque but not calculus, pocket depth, attachment loss or gingival bleeding.

 A combination of scaling and OHE will result in a more significant reduction in all periodontal parameters than scaling alone or OHE alone.

#### CHAPTER 2

#### LITERATURE REVIEW

In this chapter, the literature has been divided into five sections. The literature in the first section is concerned with the association of plaque and calculus with periodontal disease. The second section has literature which looks at the role of calculus in the aetiology of periodontal disease. The third section contains some literature on the theories of the pathogenesis of periodontal disease. The fourth and fifth sections deal with literature concerned with the effects of OHE and scaling on the periodontal status respectively.

## 2.1 Oral Hygiene Status and Periodontal Disease

Early epidemiological studies indicated a positive association between oral hygiene and periodontal disease (Emslie R.D. 1966, Sheiham A, 1970, Loe et al 1978). Loe et al 1978 found that the Sri Lankan population who had a worse oral hygiene status than the Norwegian population also had a higher severity of periodontal disease.

The aim of periodontal therapy is therefore to improve oral hygiene by removing plaque and calculus deposits from the tooth surfaces. In a clinical study, Loe et al (1965) demonstrated that gingivitis can be induced in previously healthy mouths by withdrawing oral hygiene measures and allowing plaque to accumulate. Re-institution of oral hygiene measures resulted in a

reversal of all signs of gingivitis and the gums reverted to their healthy state within a few days. This study demonstrated the importance of plaque in the development of gingivitis and the implications of the study was that had the plaque remained undisturbed for long enough, then periodontitis would have resulted.

Studies on experimental periodontal disease in animals appear to support this hypothesis. Leena S. (1985) cited a study by Lindhe et al (1973) in which they succeeded in producing *periodontitis in Beagle dogs by allowing plaque to accumulate on* their teeth for 18 months. Periodontitis did not develop in the control group of beagle dogs which received regular tooth brushing. Today, the importance of the role plaque plays in the aetiology and progression of periodontal disease is still accepted. The current belief is that periodontitis is a result of a sequential change of the bacteria found in plaque (Proceedings of Periodontology Today, 1988). The role of calculus in the aetiology of periodontal disease is however less clear.

#### 2.2 Calculus and Periodontal Disease

Calculus has for a long time been assumed to be a major aetiological factor in periodontal disease. This is mainly because of the clinical observation of an association between calculus and periodontal disease. However, this role of calculus

in periodontal disease has been questioned. It has been suggested that calculus is a result rather than a causative agent of periodontal disease (Goldman H. 1986). In addition, it has been observed in a number of epidemiological studies that the correlation of plaque to the gingival health is stronger than the correlation between calculus and gingival health (Mandel et al, 1986). A general finding of some of the studies cited in the above review (Mandel et al) is that although sites with calculus tend to have gingival inflammation, there are many more sites with gingival inflammation which had no calculus. It would therefore appear that calculus may not be a necessary factor in periodontal disease pathogenesis.

The role of calculus in periodontal disease has been thought to be indirectly through bacteria embedded in its structure or directly through toxins and antigenic substances observed to permeate the calculus. This way, calculus is believed to promote the progression of the disease apically (Goldman, 1986). However, in a study using a scanning electronic microscope, Eide et al (1983) demonstrated a mineralised layer apical to the calculus on the root surface.

This surface coating was thought to contain toxic substances. If this is the case then this surface coating may play a more important role in the apical progression of periodontal disease than calculus. Whatever role calculus plays

in the aetiology and pathogenesis of periodontal disease, its removal from the tooth surface appears to be very important in the control of the disease. This has been demonstrated in several clinical studies (Tagge et al 1975, Cerek et al 1983).

#### 2.3 Pathogenesis of Periodontal Disease

The rationale for conventional periodontal therapy is based on a model of periodontal disease in which the disease is believed to progress continuously in a linear fashion throughout the mouth. By this model, the disease is thought to progress continuously in the absence of treatment until tooth loss occurs (WHO 1978). However, this model of periodontal disease has been challenged (Socransky et al 1984, Goodson et al 1982).

Current literature suggests models of the disease in which destruction occurs in "bursts" at different sites of the mouth in a random fashion. In these models "active" and "inactive" sites occur in the same mouth. If this is true, then the rationale for treating all sites during therapy, as is done in conventional periodontal therapy, becomes obscure. Furthermore, the fact that a reversal of attachment loss has been observed to occur spontaneously without treatment in some sites (Lindhe et al 1983) appears to challenge the belief that healing of a site with periodontal disease cannot occur without treatment.

One of the reasons why periodontal therapy has been considered so important has been the belief that untreated periodontal disease causes a substantial amount of tooth loss before the age of 50 years. According to the WHO (1978), periodontal disease was said to deprive people of their teeth long before old age. However, recent studies on untreated periodontal disease indicate that the disease may be compatible with the long term retention of a functional dentition in some populations (Manji et al 1988, Baelum et al 1986, 1988).

# 2.4 The Effect of Oral Hygiene Measures alone on the Periodontal Status

The effect of oral hygiene measures in the periodontal status depends on the patients' compliance (Loos, 1988) which in turn depends on a number of social and psychological factors (Woodwal R.I. 1984). Oral hygiene measures have been shown to be effective in controlling gingivitis (Loe, 1965). The effect of oral hygiene measures alone on periodontitis is not extensively studied. In most of the studies in literature this effect has only been studied over short period of 2-6 months. There appears to be no studies on the long term effect of oral hygiene measures alone on untreated periodontal therapy.

The effects of oral hygiene measures on the periodontal status in untreated periodontal disease has been studied in a number of clinical experiments (Tagge et al 1973, Badersten et al
1984, Loos et al 1988, Cerek et al 1983). The general finding is that oral hygiene measures can reduce the plaque score significantly. However, the improvement on the bleeding score and pocket depth is limited. Minimal or no effect is observed on the attachment level.

The improvement of periodontal parameters appear to depend on the initial depth of the pockets. For plaque, lower plaque scores are obtained in shallow pockets  $\leq 3$  mm than in deep pockets. The bleeding score has a greater improvement in the shallower pockets. For the pocket depth, the greatest improvement occurs in the deep pockets (24 mm).

The improvement resulting from oral hygiene measures on untreated periodontal disease has been maintained for 3 months (Loos et al 1988, Badersten et al 1981) and 6 months (Cerek et al 1983). However, Cerek et al (1983) observed a deterioration of the attachment level after 8 months or oral hygiene measures alone. It would therefore seem that oral hygiene measures alone cannot control the progression of periodontal disease on a long term basis.

The limitation of oral hygiene measures on the control of periodontal disease has been attributed to its ineffectiveness in altering the subgingival microflora. Loos et al (1988) demonstrated that even when oral hygiene measures were complied

with, an alteration of the composition of the subgingival microflora did not occur.

#### 2.5 The Effect of Scaling on the Periodontal Status

The effectiveness of scaling in improving the periodontal status is well documented in literature. Many of these studies are usually designed to compare the effects of surgical and nonsurgical treatment modalities. The general finding of these studies is that both surgical and non-surgical periodontal therapies are equally effective in improving and maintaining the periodontal status on a short-term (Lindhe et al 1982) and on a long-term basis (Ramjford et al 1973). Other studies compare the effects of scaling and of oral hygiene measures alone on the periodontal status. These have indicated that the most marked improvement on the periodontal status occurs only after scaling (Badersten 1984, Cerek 1983, Tagge et al 1973).

Generally, scaling has been found to result in the improvement of all periodontal parameters. The magnitude of its effect appears to depend on the initial depth of the pockets. The reduction of the gingival score is more marked on the shallo pockets ( $\leq 3$  mm). The greatest improvement of the pocket depth an attachment loss occurs in the deep pockets (Lindhe 1987). A deterioration of the pocket depth and attachment level after scaling has been observed in shallow pockets in a number of studies (Lindhe et al 1982, Philstrom et al 1981, Badersten et al 1981).

The effectiveness of periodontal therapy also appears to differ with the different types of teeth. Non-molar teeth respond better to therapy than molar teeth (Lindhe et al 1982). Philstrom et al (1984) in a longitudinal study found that periodontal therapy significantly improved the periodontal status of both molar and non-molar teeth. There was a greater reduction in the pocket depth and attachment level in the pockets initially 4-6 mm in the non-molar teeth. However, no difference was observed in pockets 27 mm.

Scaling has been found to produce a significant improvement of the periodontal status within a few weeks after therapy. Proye et al (1982) demonstrated a significant improvement of the pocket depth within 4 weeks after a single episode of root planning. This was attributed to be a result of gum recession and partly due to a gain in the attachment level. The bleeding score also decreases significantly within the same period.

The improvement on the attachment level obtained after therapy has been maintained unchanged or with little change for three years (Lindhe et al 1987) and over (Ramjford 1973). The most important factor in the long-term maintenance of the effects obtained after periodontal therapy is good post-treatment plaque

control. Where this is not adequate, a deterioration occurs. Pruthii V.K. (1986) in his review cited a histological study by Stahl et al in which the inflammatory infiltrate was observed to return to pre-treatment level after 52-60 days. This deterioration was attributed to the fact that no post-treatment plaque control measures were taken. Studies that have compared groups with good and poor post-therapy plaque control have found that in those groups with poor control, periodontal disease continued to progress while in those groups with good plaque control the disease was arrested, (Axlesson and Lindhe 1978, Soumi et al 1969). In a retrospective study on periodontal treatment without maintenance therapy, Becker W. et al (1984), stated that periodontal therapy without maintenance is of little value.

There were no randomised clinical studies in literature comparing the effects of scaling with those of oral hygiene measure alone on the periodontal status of populations with untreated periodontal disease.

#### CHAPTER 3

#### METHODS AND MATERIALS

This chapter contains a description of the study area, the study design, the study population, the sample size determination, the selection and randomization procedures in the first section. The instruments used in data collection, methods and instruments used in the oral examinations, indicators of periodontal status and their measurement are described in the second part. The third section is on treatment alternatives and instruments used for treatment. Finally, tables showing the time schedule, results of the intra-observer reliability and an outline of the analysis in the study are displayed.

#### 3.1 Study Area

This study was conducted in 2 factories situated in the industrial area in Nairobi, the capital city of Kenya. The industrial area has about 2,000 factories lying approximately 5 Kms East of the city centre and stretching over an area of 200 Kms.

The two factories selected for this study were chosen from a list of "Jarge" factories (i.e. with 200-300 factory workers) obtained from the Factories Inspectorate Department. These factories were selected on basis of convenience in terms of distance, large number of workers and availability of suitable

health care facilities at the work site. One of the factories chosen specializes in production of wheat flour and the other whole maize meal. Each factory had about 250-300 male and female workers excluding the managerial staff. The estimated male to female ratio was 3:1. Each factory had a well equipped dispensary where the examinations for this study were carried out.

#### 3.2 Study Design

The study was a randomized clinical trial. A group of factory workers were selected and randomized into four treatment groups. Examinations were performed at baseline and six months following treatment.

The design of the study was factorial as illustrated below.

	YES	NO	
NO	n <sub>1</sub>	$n_2$	$n_1 + n_2$
YES	n <sub>3</sub>	n <sub>4</sub>	$n_3 + n_4$
TOTAL	$n_1 + n_3$	n <sub>2</sub> + n <sub>4</sub>	Ν

#### ORAL HYGIENE EDUCATION

SCALING

This design facilitated the computation of a combination of scaling and O.H.E. It also offered an advantage in that a smaller study sample was needed than would have been with other designs.

#### 3.3 Study Population

This comprised of all male factory workers aged 30-50 years from two milling factories; 147 from the wheat milling factory and 110 from the maize meal factory. Non unionized workers (management staff) were excluded from the study as they were considered to belong to a higher socio-economic status with better oral hygiene. They were more likely to have had scaling. A preliminary review indicated that non-unionized individuals in the study had minimal dental treatment.

#### 3.4 Sample Size Determination

In order to demonstrate a difference between 2 groups of at least half a standard deviation with a significance (one sided) of 0.05 and a power of 0.80, it was found that 50 individuals in each group were required.

The equation used was:

n = 2 (k(α) + k(1 - β) σ<sup>2</sup> d<sup>2</sup> where n = required sample size per group σ = the standard deviation in each individual observation α = the required level of significance β = the power of the test d = the required difference between the 2 means where the difference is at least half a standard deviation.

#### 3.5 Sample Selection and Randomization

Two lists of all male factory workers aged between 30-50 years were made, one for each factory. The lists included 147 names from the first factory and 110 names from the second factory. From each list, the workers were called one at a time to the factory dispensary. The purpose of the study was explained to each worker after which a verbal consent of their willingness to participate in the study was obtained. Those who were not willing to participate were immediately eliminated from the study. Those who were willing to participate were screened to further determine whether they were eligible for inclusion or not. A list of the names of all eligible subjects was then made to act as the sampling frame from which a total of 101 subjects were randomly selected; 51 from and 50 from the first and second factory, respectfully.

Using the sampling frame, subjects were randomized into 4 treatment groups of approximately equal sizes using a table of random numbers. There were 26, 26, 25 and 24 in groups 1, 2, 3, and 4 respectively.

#### 3.6 Eligibility Criteria

#### 3.6.1 Inclusion criteria:

- 1. Males 30 50 years.
- 2. non-unionized (non-management level staff)
- 3. Willingness to participate in the study.

#### 3.6.2 Exclusion Criteria

- Presence of pockets ≥4 mm on more than 4 teeth. It was felt that pockets ≥4 mm could not be adequately cleaned without surgery.
- Subjects with medical conditions that could have affected periodontal status e.g. diabetes, epilepsy.
- Subjects who had periodontal surgery or scaling over the previous 5 year period.
- 4. Any worker who was not available at the time of the preliminary examination (prior to study subject selection and randomization) for any reason.

#### 3.7 Ethical considerations

- 1. An informed verbal consent for inclusion into the study was obtained from all the participants.
- 2. Pain relieving treatment was given as required. Dental advice among subjects who were randomized to non OHE groups was only offered on request. Even then, it was deliberately kept brief and strictly to the issue in question.

#### 3.8 Data Collection

The following information was collected on a questionnaire

by a research assistant (appendix 1): name, age, brushing habits, type of brush used, smoking and drinking habits. Clinical examinations to determine the plaque, calculus, pocket depth, attachment loss and gingival bleeding scores were carried out before treatment (baseline examination) and 6 months after treatment (final examination). These examinations were done by the principal investigator and recorded on an examination sheet (Appendix II) by a research assistant. For each participant, the results at the baseline and final examination were recorded on separate examination sheets.

#### 3.9 Oral Examination

Oral examinations were carried out with the aid of dental mouth mirrors and periodontal probes. These probes had markings at 3 mm, 5 mm and 8 mm. A lamp with a 40 watt bulb was used to illuminate the oral cavity during these examinations. The same instruments were used during the baseline and final examinations.

Both the baseline and final examinations were carried out by the principle investigator at the factory dispensaries. The baseline examinations were carried out before the treatments were done and the final examinations were carried out 6 months following treatment. The examinations were carried out using a periodontal probe to measure plaque, calculus, gingival bleeding *pocket depth and loss of attachment scores in that order. These* measurements were taken from the buccal, lingual, mesiobuccal and distobuccal aspects of all the teeth except the third molars.

Plaque was measured visually and by running a probe on the tooth surface. Calculus was recorded depending on the type of calculus that was present on the particular surface. Presence of subgingival culculus was identified using a periodontal probe and gingival bleeding by running the probe gently in the depth of the gingival sulcus.

Measurement of pocket depth and attachment loss from the buccal and lingual surfaces were taken along the flat surfaces of the molar teeth and within a distance of 2 mm on either side of the midline for the non-molar teeth, the highest measurements were recorded. The disto-buccal and mesio-buccal measurements were taken as close to the contact point as possible while keeping the probe parallel to the long axis of the tooth. The pocket depth and attachment loss were measured to the nearest millimetre using a periodontal probe. In many sites, the position of the cementoenamel junction was estimated because it was difficult to probe it mainly because of the presence of calculus.

Tooth mobility and missing teeth were recorded. Missing teeth were recorded according to the causes of tooth loss. This was determined by inquiring from the patient as to the reason for tooth extraction. If the reason for a tooth extraction was presence of a hole, this was recorded as missing due to caries, if it was extracted because of mobility then it was recorded as

missing due to periodontal disease. Determination of missing teeth due to traditional extraction was made from an interview and examination (nearly always involved loss of lower anterior teeth only). It involved the lower anterior teeth only. Teeth missing due to other reasons included teeth that had been lost due to trauma and teeth that had not erupted.

#### 3.10 Measurement Criteria

Plaque Score (Silness and Loe 1964)

0 = no plaque

1 = no visible plaque but plaque present on probing

2 = plaque visible without probing

3 = abundant plaque covering more than 1/2 the tooth surface or filling the proximal space.

Gingival inflammation (slight modification of index by Silness and Loe 1967).

0 = no bleeding

1 = slight bleeding

2 = profuse bleeding

3 = exudation of pus

Calculus Score (Bjorn and Loe 1967)

- 0 = no calculus
- 1 = supra gingival calculus
- 2 = subgingival calculus including dark staining calculus deposits in cases of gum recession
- 3 = both supra and subgingival calculus on the same surface

#### Pocket depth

This was measured with periodontal probe from the margin of the gingiva to the depth of the pocket. The measurement was taken to the nearest millimetre. Any pocket 28 mm was scored as "8".

#### Loss of attachment

This was measured with a periodontal probe from the cementoenamel junction to the depth of the pocket. The measurement was taken to the nearest millimetre.

#### Tooth Mobility

- 0 = no mobility
- 1 = horizontal mobility <2 mm
- 2 = horizontal mobility >2 mm plus vertical mobility

#### Causes of tooth loss

- 0 = tooth present
- 1 = missing due to other reasons
- 2 = missing due to traditional extraction
- 3 = missing due to caries
- 4 = missing due to periodontitis

#### 3.11 Treatments

These were carried out within 3 to 4 months after the baseline examinations. This delay was due to a problem in obtaining the scaling equipment. All treatments were administered by the same investigator to ensure consistency in therapy.

#### Treatment groups:

- 1. Scaling only
- 2. Scaling and O.H.E.
- 3. Oral hygiene education only
- 4. No treatment (Control)

#### 3.11.1 Scaling

This was done using an sonic scaler with a Mijet ( Meut model, no.T(XL033) at one sitting without local anaesthesia. Scaling of each individual took on average two and a half to three hours. Polishing was done on only a single occasion from 3 days to 2 weeks after scaling using a rubber cap or polishing brush using pumice mixed with prophylaxis paste. During polishing, any visible calculus was removed.

#### 3.11.2 Oral Hygiene Education

Oral hygiene instructions were given individually to each participant in the groups receiving O.H.E. In the group that received a combination of scaling and OHE, the latter was given immediately after the scaling. A guide sheet (Appendix III) was used to standardize the instructions. No special brushing technique was taught.

No reinforcement of OHE was given for the rest of the study.

			,		
	GP1	GP2	GP3	GP4	Month the procedure was done
Baseline Examination	+	+	+	+	Mid December to mid February 1988
Scaling and Polishing	÷	÷	_	-	Mid March to mid May 1989
Oral Hygiene Education	_	+	+		May
Final Examination	+	+	+	+	August to september 1989

#### 3.12 Schedule of the Study

#### 3.13 INTRA-OBSERVER RELIABILITY

During the baseline and final examinations, every tenth worker examined had a second examination performed, on a randomly selected quadrant, 2 - 3 hours after the initial examination. The repeat examinations were done for calculus, pocket depth and attachment loss to determine the intra-observer reliability for these parameters. Repeat examinations for plaque and gingival bleeding were not feasible because the first examination introduced changes that interfered with the initial periodontal findings. For calculus, the simple agreement based on scores between the 1st and 2nd readings at the baseline and final examinations was 88% (see appendix IV).

For the pocket depth at the baseline examination, 100% of the 1st and 2nd readings agreed within  $\pm 2$  mm, 99% of the reading agreed within  $\pm 1$  mm (see appendix IV). At the final examination all the 1st and 2nd readings agreed within  $\pm 3$  mm. However less than 1% of the sites differed by  $\pm 3$  mm. The majority (97%) agreed within  $\pm 1$  mm.

For the attachment level at the baseline examination, 100% of the 1st and 2nd readings agreed within  $\pm 3$  mm. Majority of the readings (95%) agreed within  $\pm 1$  mm. Only less than 1% differed by  $\pm 3$  mm. At the final examination 100% of the 1st and 2nd readings agreed within  $\pm 2$  mm. However majority of the readings (97%) agreed with  $\pm 1$  mm (see appendix IV).

#### 3.14 ANALYSIS

Only data from participants who were present at both the baseline and final examinations were used in the analysis. Those subjects who were supposed to have scaling or OHE or both but did not, for logistic reasons, but who were available at both examinations were analyzed in the "no treatment" group. Analysis of baseline characteristics suggested that the subgroup did not systematically differ from the other subjects in the randomized groups.

Analysis was done using an SPSS-PC statistical package. A comparison of the groups at the baseline examination for each parameter was done using an analysis of variance (ANOVA) on the number of sites with the different scores. A paired student's ttest was used to examine the differences that occurred within each group between the baseline and final examination. A comparison of the groups at the final examination was done by ANOVA on the differences that occurred within the groups in the number of sites with the different scores. A student Newman Keul's test was used on the results that indicated a significant difference to determine the source/s of differences.

To determine the effects of scaling on the periodontal disease groups 1+2 were combined into the "scaled" group which was then compared with the "non scaled" group (group 3+4). To determine the effects of OHE, group 2 and 3 were combined into the "OHE" group which was then compared with the "non OHE" group (group 1+4).

#### CHAPTER 4

#### RESULTS

This chapter is divided into five major sections. Section 1 contains the results of the general characteristics of the population. These include the mean age and age range, the general tooth status of the participants. Section two shows the distribution of the participants after randomization, at baseline and at the final examination. The third section contains results of the comparison of the groups at the baseline examination. The third section contains the results of the differences that occurred within the groups between the baseline and final examinations. Comparisons between the "scaled" and "non scaled" groups, the "OHE" and "non OHE groups", and the scaling only and scaling combined with OHE groups of the differences that occurred within them after treatment is contained in the fifth section.

#### 4.1 GENERAL CHARACTERISTICS OF THE STUDY POPULATION

The study population comprised of 101 men. Their mean age was 39.5 years with a range of 30 to 50 years. Eighty-four percent of these workers claimed to brush their teeth at least once daily. Majority of these workers, (81.2%) brushed their teeth with a conventional toothbrush, while 17.8% used a chewing stick (mswaki). None of the participants used an anti-tartar toothpaste.

The dental treatment which had been received by the

participants in this study prior to the study was limited to dental extractions. There were no fillings in this study group and only one worker had received periodontal therapy ten years previously.

The total number of mobile teeth was 36.i.e roughly one mobile tooth in every third participant. However most of the mobile teeth were observed to occur in a few individuals only while in majority of the participants no mobility was observed.

The total number of missing teeth among study participants was 123. Of these, 77 (63%) were lost due to caries, 24 (20%) due to traditional extraction and 20 (16%) due to periodontal disease.

# 4.2 DISTRIBUTION OF STUDY SUBJECTS IN THE TREATMENT GROUPS

### Table 1: Frequency distribution of subjects in the 4

trea	tment	groups
	supported by the same second second second	the second secon

Treatment	After randomiza- tion	During the baseline analysis	During the final analysis
Scaling only	26	26	24
Scaling + OHE	26	17	15
OHE	25	19	15
No treatment	24	39	25
TOTAL	101	101	79

After randomization the 4 groups had approximately equal numbers of subjects. During the analysis of the baseline results, the numbers in the groups changed following the decision to analyze results of subjects who were randomized to groups 1, 2 and 3, and who did not receive either scaling or OHE, with group 4 subjects. The number of individuals in group 1 (scaling only) remained the same. The group that had a combination of scaling and OHE and the OHE only group experienced a decrease in numbers while the number in the no treatment group increased. During the final examination, the total number of participants had decreased by 22. The number of subjects in each of the groups at baseline and final analysis are shown in table 1. A total of 2, 2, 4 and 14 subjects dropped out of groups 1, 2, 3 and 4 before the final examination respectfully.

Treatment	% of Participants			
	at random- ization	at baseline analysis	at final analysis	
Scaling only	25.7	25,7	30.4	
Scaling + OHE	25.7	16.8	19.0	
ОНЕ	24.8	18.8	19.0	
No treatment	23.8	38,6	31.6	
TOTAL	100	100	100	

#### Table 2: Percent distribution of subjects by treatment groups

#### 4.3 COMPARISON OF THE GROUPS AT THE BASELINE EXAMINATION

#### 4.3.1 Plaque

In all the groups the mean number of sites that were calculus free was less than 30. Majority (48-54) of the sites exhibiting plaque had little amounts of plaque (score 1). The mean number of sites with plaque score 2 in the 4 groups was between 30-33. The mean number of sites with great amounts of plaque (score 3) was slightly higher (2) in the OHE group than in the other three groups where they were less than one. Generally, the distribution of the various plaque scores is quite evenly distributed in the four groups.

> MEDICAL LIBRARY UNIVERSITY OF NAIROBI P. O. Box 19676 NAIROBI

Treatmont	Score				
Ileatment	0 .	1	2	3	
Scaling only	26.2	47.5	30.8	0.84	
	(21.4)	(20.2)	(24.8)	(2.1)	
Scaling + OHE	21.8	53.7	29.9	0.88	
	(20.5)	(14.0)	(20.3)	(2.0)	
OHE	25.9	46.3	33,3	2.47	
	(22.3)	(21.3)	(27,9)	(6.1)	
No Treatment	23.0	51.3	32.8	0.72	
	(20.9)	(15.1)	(24.2)	(1.9)	

#### Table 3a: <u>Mean Number of Sites with Plaque Score 0,1,2,3 and</u> standard deviations at baseline by treatment groups.

The standard deviations are indicated in parentheses.

Table 3b:F and P Values for Plaque scoresamong treatment Groups at Baseline

Score	F-Value	P-Value
0	0.23	0.87
1	0.77	0.52
2	0.10	0.96
3	1.44	0.24

o Significant differences were observed among the groups.

#### 4.3.2 Calculus

Table (4a) shows that the mean number of sites in the four groups that exhibited calculus was between 50 and 60. The mean number of sites with supragingival calculus (score 1) in the four groups was 6-8. Most (43-51) of the sites had subgingival calculus.Only a few sites (a mean of less than 5 sites in all the four groups) had supragingival and subgingival calculus (score 3) on the same surface.

	Score				
Treatment	0	1	2	3	
Scaling only	48.7	7.8	44.9	4.2	
	(28.2)	(5.5)	(26.5)	(11.8)	
Scaling + OHE	51.8	5.6	45.9	1.6	
	(24.1)	(4.7)	(23.6)	(2.1)	
OHE	56.8	6,3	42.5	2.3	
	(25.0)	(3,8)	(25.4)	(2.9)	
No treatment	49.0	6.0	51.1	1.8	
	(23.4)	(2.4)	(22.1)	(2.4)	

Table 4a: The Mean number of sites with calculus score 0,1,2,3 at baseline by treatment groups.

The mean number of sites with the various scores was similar in all the groups and no significant differences were observed (tables 4, 4b).

Score	F-Value	P-Value
0	0.50	0.68
1	1.12	0.34
2	0.67	0.57
3	0.51	0.67

Table 4b:The F and P Values for the Differences in CalculusBetween the 4 Groups at Baseline

#### 4.3.3 Gingivitis

The groups on table 5a below show that the mean number of sites having gingivitis in the four group was between 62-66 sites; majority (57-61) had mild bleeding (score 1). The mean number of sites with gingivitis score 2 was between 3-6. The mean number of sites with pus (score 3) was less than 1 in all the groups.

Table 5a:	Mean Number of	Sites with G	ingivitis Sco	ore
	0,1,2,3 and	the standard	deviations a	at baseline by
	treatment	groups		

Treatmont	Score				
Ileatment	0	<u>1</u>	2	3	
Scaling only	44.0	56.7	4.7	0.04	
	(25.5)	(24.6)	(9.9)	(0.20)	
Scaling + OHE	39.1	61.0	5.8	0.12	
	(23.0)	(21.9)	(5.9)	(0.5)	
OHE	45.5	57.4	5.0	0.0	
	(21.4)	(18.8)	(5.2)	(0.0)	
No Treatment	45.0	59.4	3.4	0.1	
	(21.4)	(23.7)	(4.0)	(0.4)	

Score	F-Value	P-Value
0	0.28	0,84
1	0,16	0,93
2	0.65	0.58
3	0,62	0.60

Table 5b ;F and P Values for the Differences inGingivitis Between the Treatment groups at<br/>Baseline.

There were no significant differences between the groups in the mean number of sites with the various scores for gingivitis (tables 5a ,5b).

#### 4.3.4 Pocket Depth

Majority of the sites had shallow pockets. The mean number of sites with shallow pockets in the four groups was between 101-103. Deep pockets were few and the mean number of sites with deep pockets was 5 or less in all the groups.

Table 6a: Mean Number of Sites with Shallow ( $\leq 3$  mm) and Deep ( $\geq 4$  mm) Pockets and the standard deviations at Baseline.

	Score			
Treatment	≤3 mm	≥4 mm		
Scaling only	101.7 (9.8)	3.7 (4.6)		
Scaling + OHE	101.0 (8.2)	5,1 (5,4)		
OHE	103.4 (3.8)	4.7 (5.5)		
No Treatment	103.3 (7.8)	4.6 (5.1)		

Table 6b:F and P Values for the Differences inPocket Depth Between the Groups at Baseline

George	Differences		
SCOLE	F-Value	P-Value	
≤3 mm	0.43	0.74	
≥4 mm	0.33	0.80	

There were no significant differences between the groups either in the mean number of shallow pockets or in the mean number of deep pockets (tables 6a, 6b).

#### 4.3.5 Attachment Level

Table 7a below indicates that on average the individuals in the four groups had lost some attachment on most (over 70) of their sites. Most ( 66-67 ) of the sites had lost 1-3 mm of attachment. The mean number of sites with attachment loss of 4 mm or more was less than 8 sites in all the groups. The mean number of sites without attachment loss was between 33-35 in the 4 groups.

Table 7a:	Mean	Number	of	Sites	with	Attachment	Loss	0	mm,	1 - 3	mm,
	≥4 mr	n and th	ne	standar	rd dev	viations					

	Score				
Treatment	0	1-3	.≥4 mm		
Scaling only	33.2	66.7	5,5		
	(27.0)	(21.5)	(12,9)		
Scaling + OHE	35.4	66.8	4,0		
	(23.8)	(23.7)	(4,0)		
OHE	34.4	66.3	7.2		
	(22.9)	(21.4)	(11.9)		
No Treatment	33.7	65.5	7.6		
	(27.0)	(22.5)	(11.2)		

There were no significant differences between the groups in any of the attachment levels (tables 7a, 7b).

Score (mm)	F-Value	P-Value
0	0.28	0.99
1-3	0,001	1.0
<u>≥</u> 4	0.52	0.67

Table 7b:F and P Values for Differences in Attachment LevelBetween the Groups at Baseline

## 4.4 THE DIFFERENCES THAT OCCURRED WITHIN THE GROUPS BETWEEN THE BASELINE AND FINAL EXAMINATION

The differences between the two examinations were obtained by subtracting the baseline results from the final results. A paired t-test was used to determine whether the differences that occurred were significant.

#### 4.4.1 Plaque

The mean number of plaque free sites increased in all the groups. The highest increase (25) was in the scaling combined with OHE group and the smallest (9) was in the 'no treatment' group. In the scaling only, scaling combined with OHE and the OHE groups, the mean number of sites with plaque score 1,2,3 decreased. In the 'no treatment' group, the mean number of sites with plaque score 1 decreased while the mean number of sites with plaque score 2 and 3 increased by 6 and 1 sites respectively.

ngel	Score				
Treatment	0	1	2		
Scaling only	18.17	-5.54	-12.46	-0.13	
	(12.21)	(22.35)	(22.07)	(2.98)	
Scaling + OHE	25.33	-13.87	-11,60	-0.20	
	(23.56)	(17.81)	(20,20)	(18.94)	
OHE only	17.89	-6.00	-10.60	-0.93	
	(16.23)	(15.15)	(19.20)	(5.36)	
No Treatment	9.25	-16.4	6,25	-0.92	
	(19.47)	(18.94)	(22,05)	(3.13)	

# Table 8:Means of the Differences in the number of sites with<br/>plaque score 0,1,2,3 in the treatment groups

An analysis of the differences on table 8 indicated that there was a significant increase in the number of plaque free sites in all the groups at the final examination (t=3.93, p<0.001; t=3.60, p < 0.01; t=4.26, p<0.01;t=2.45, p<0.05) for groups 1,2,3 and 4 respectively).

#### 4.4.2 Calculus

The mean number of calculus free sites increased greatly in the scaled groups by 35-37 sites while in the groups that did not receive scaling, an increase by 4 sites occurred in both groups. The mean number of sites with supragingival calculus ( score 1) changed by 4 sites in the scaling only group, while in the other three groups there was little change. The greatest decrease in calculus occurred in the subgingival calculus.

e mean number of sites with subgingival calculus decreased by -33 sites in the scaled group. while in the groups that did not reive scaling, an increase of 6 sites occurred in both groups. • biggest decrease (5) in the mean number of sites with score 3 [culus occurred in the scaling only group.

ble 9:

#### Means of differences in the

#### number of sites with Calculus score 0,1,2,3 in the Treatment groups

	Score					
Treatment	0	1	2	3 .		
Scaling only	39.17	-3.92	-30.92	-4.50		
	(20.56)	(5.16)	(19.50)	(12.29)		
Scaling + OHE	35.07	-0.46	-32.93	-1.60		
	(18.4)	(3.96)	(19.04)	(2.38)		
OHE only	-4.07	-0.67	5.93	-0.80		
	(8.50)	(2.47)	(7.62)	(2.96)		
No Treatment	-3.88	-1.12	5.60	-0.32		
	(9.89)	(3.03)	(9.25)	(1.65)		

n analysis of the differences on table 9 indicated that there as a significant increase (t=9.33, p<0.001 ;t=7.37, p<0.001 in roups 1 and 2 respectively) in the number of calculus free sites n the groups that received scaling. The number of sites with upragingival calculus decreased significantly (t=3.70, p<0.01) n group 1 (scaling only). There was no significant change in he other groups. The number of sites with subgingival calculus decreased significantly (t=7.77, p<0.001; t=6.69, p<0.001) in groups 1 (scaling only) and 2 (scaling combined with OHE) respectively, while in groups 3 (OHE only) and 4 (no treatment), a significant increase occurred (t=3.03, p<0.01; t=2.34, p<0.05, respectively). There was no significant change in the number of sites with supragingival and subgingival calculus on the same surface (score 3) in any of the groups.

#### 4.4.3 Gingivitis

The mean number of gingivitis free sites increased in all the groups and especially in the scaled groups. The mean number of sites with gingivitis score 1,2 decreased in all the groups with the scaled groups having the greatest decrease. There was little change in score 3 gingivitis in all the groups.

# Table 10:Means of differences in the number of sites with<br/>Gingivitis score 1,2,3 in the Treatment groups.

	Score				
Treatment	0	1	2	3	
Scaling only	19.29	-15.08	-4.13	-0.042	
	(23.62)	(22.04)	(10.00)	(0.20)	
Scaling + OHE	25.40	-19.20	-6.00	-0.13	
	(26.78)	(25.34)	(6.00)	(0.52)	
OHE	8.33	-5.00	-3.00	0.0	
	(11.79)	(12.75)	(2.88)	(0.0)	
No Treatment	6.12	-5.16	-0.64	0.0	
	(20.77)	(19.92)	(2.53)	(0.29)	

An analysis of the differences in table 10 indicated that the mean number of gingivitis free sites increased significantly (t=4.00, p<0.001; t=3.67, p<0.01) in groups 1 and 2 respectively. In groups 3 and 4 no significant changes occurred.

#### 4.4.4 Pocket depth

The mean number of sites with shallow and deep pockets changed by 2-3 sites only in the scaled groups while in the groups that did not receive scaling these hardly changed.

# Table 11:Means of differences in the number of sites with<br/>shallow (≤ 3 mm) and deep (≥4 mm) Pockets in the<br/>Treatment Groups

	Score			
Treatment	⊻ 3 mm	≥4 mm		
Scaling only	2.29 (3.43)	-2.17(3.06)		
Scaling + OHE	2.60 (3.40)	-2.60 (3.48)		
OHE	0.47 (2.59)	-0.27 (2.31)		
No Treatment	0.72 (3.12)	-0,44 (3,02)		

An analysis of the differences on table 11 indicated that the number of shallow pockets increased significantly (t=3.27, p<0.01; t=2.95, p<0.01) in groups 1 and 2 respectively. The number of deep pockets (24 mm) decreased significantly in these groups (t=3.50, p<0.01; t=2.89, p<0.05 respectively). There was no significant change in groups 3 and 4.

#### 4.4.5 Attachment level

The mean number of sites with attachment loss 0 mm increased in all the groups. The mean number of sites with attachment loss 1-3 mm decreased in all the groups by approximately the same number of sites as those that increased in the attachment loss 0 mm. The mean number of sites with an attachment loss  $\geq$ 4 mm decreased by 1-2 sites in the groups that received scaling and hardly changed in the groups that did not receive scaling.

Table 12: Means of the Differences in the mean number of sites with Attachment Level 0 mm, 1-3 mm, and 24 mm in the Treatment Groups

	Score			
Treatment	0 mm	1-3 mm	.≥4 mm	
Scaling only	37,08	-35.67	-1.25	
	(18,58)	(18.52)	(3.37)	
Scaling + OHE	39.67	-37.93	-1.73	
	(18.36)	(18.85)	(2.58)	
OHE	31.40	-31.47	0.04	
	(18.37)	(18.60)	(3.40)	
No Treatment	26.68	-26.80	0.32	
	(11.14)	(12.14)	(3.82)	

An analysis of the differences in table 12 indicated that the number of sites with attachment loss 0 mm increased significantly in all the four groups (t=9.78, p<0.001; t=8.37, p<0.001; t=6.95, p<0.001; t=11.96, p<0.001 for groups 1,2,3 and 4 respectively).

The number of sites with attachment loss 1-3 mm decreased significantly (t=9.64, p<0.001; t=7.79, p<0.001; t=6.54, p<0.001; t=10.98, p<0.001 for groups 1,2,3 and 4 respectively). The number of sites with attachment loss  $\ge 4$  mm decreased significantly (t=2.58, p<0.05) in group 2 only. There was no significant change in groups 1, 3 and 4.

# 4.5 BETWEEN GROUPS COMPARISON OF THE DIFFERENCES THAT OCCURRED WITHIN THEM AFTER TREATMENT

#### 4.5.1 PLAQUE

## 4.5.1.1 Comparison of the Differences that Occurred in the 4 Groups for Plague

The means of the differences in plaque scores in the four groups are shown in table 8. The number of plaque free sites increased in all the four groups. The number of sites with plaque score 1,2,3 decreased in the scaling only , the scaling combined with OHE and the OHE only groups. In the no treatment group, the number of sites with plaque score 1 decreased while those with plaque score 1 and 2 increased slightly.

No significant differences occurred between the four groups for plaque score 0, 1, 3. A significant difference occurred in the change that occurred in the number of sites with plaque score 2 (F= 4.24; p= 0.008). A student Newman-Keuls test indicated that this difference occurred between group 4 (no treatment) and the other three groups (p<0.05, q=3.70).
#### 4.5.1.2 The Effects of Scaling on Plaque

This was determined by comparing the differences that occurred within the "scaled" group (i.e. group 1+2) with that which occurred within the "non scaled" group (i.e. group 3+4). (N.B. the same was done for the other parameters).

The number of plaque free sites increased in both the scaled and non scaled groups (table 13). The number of sites with plaque score 1 decreased in both groups. In the "scaled" group, the mean number of sites with plaque score 2 decreased by 12 sites while in the "non scaled" groups there was hardly any change. A significant difference between the two groups occurred in plaque score 2 only. There was little change in plaque score 3. The "scaled" group had a significantly (F=2.15; P=0.05) greater decrease in the number of sites with plaque score 2 than the "non scaled" group. There were no significant differences between the groups in the change that occurred in the number of sites with plaque score 0,1,3.

Treatment	Score					
Group	0	1	2	3		
Scaled (1+2)	20.92 (25.79)	-8.74 (20.68)	-12.13 (21.04)	-0.15 (2.57)		
Non Scaled (3+4)	$12.65 \\ (18.56)$	-12.65 (18.18)	0.1(22.40)	0.23 (4.15)		

#### Table 13: Means of differences in plaque score 0,1,2,3 in the "scaled" and "non scaled" groups.

#### 4.5.1.3 The Effects of OHE on Plaque

This was determined by comparing the differences that occurred within the "OHE" group (i.e. group 3+4) with those that occurred within the "no OHE" group. (N.B. the same was done for the other parameters).

The number of plaque free sites increased in both the "OHE" group and the "non OHE" group. The number of sites with plaque score 1 and 2 decreased in both group. Little change occurred in plaque score 3 in both the groups.

There were no significant differences between the "OHE" group and the "no OHE" group in the change that occurred in the number of sites with any of the plaque scores.

		Score		
Treatment Group	0	1	2	3
'OHE' (2+3)	21.60	-9.93	-11.10	-0.57
	(20.24)	(16.41)	(19.28)	(3.95)
'No OHE'	13.74	-11.20	-2.78	0.41
(1+4)	(23.74)	(21.21)	(23.84)	(3.0)

## Table 14:The means of the differences in plaque score0,1,2,3 in the "OHE" and "no OHE" groups

## 4.5.1.4 Comparison of the Effects of Scaling only with Scaling Combined with OHE on Plaque

The number of plaque free sites increased in the both the scaled groups and in the group that did not receive scaling. This increase was greatest in the scaling combined with OHE group (table 15). The number of sites with plaque score 1,2 and 3 decreased in both the groups that received scaling. The decrease in the mean number of sites with plaque score 1 was higher (14) in the scaling combined with OHE than in the scaling only group where the decrease involved 6 sites.

There were no significant differences between the group that received scaling only and that which received scaling combined with OHE in the change that occurred in the number of sites with any of the plaque scores.

Tables 15:	Means of	the	e Differ	ences	in	Plac	rue	Scor	re	
	0,1,2,3	in	scaled	(group	) 1	and	2)	and	the	"non-
	scale	d" (	3 and 4	) grou	DS					

	Score	ć		
Treatment Group	0	1	2	3
Scaling only	18.17	-5.54	-12.46	-0.13
	(27.21)	(22.37)	(22.07)	(2.98)
Scaling + OHE	25.33	-13.87	-11.60	-0.20
	(23.56)	(17.18)	(20.02)	(1.82)
Non-Scaled	12.65	-12.65	0.10	0.23
	(18.56)	(18.18)	(22.40)	(4.15)

## 4.5.1.5 Comparison of the Groups that Received OHE only, Scaling Combined with OHE and the "No OHE" Group for plaque

The mean number of plaque free sites increased in all the three groups (table 16). The increase was highest (25) in the scaling combined with OHE group. The number of sites with plaque score 1 and 2 decreased in all the groups. There was little change in plaque score 3.

There were no significant differences between the group that received OHE only (group 3), the group that received scaling combined with OHE (group 2) and the "non OHE" group (group 1+4)in the change that occurred in the number of sites with any of the plaque scores.

Table 16:	Means of	differenc	ces in p	olaque s	score			
	0,1,2,3	in group	with OF	HE (2,3)	and	without	OHE	(1, 4)

	Score	2		e.
Treatment Group	0	1	2	3
OHE only	17.87	-6.00	-10.00	-0.20
	(16.23)	(15.15)	(19.20)	(1.82)
Scaling + OHE	25.33	-13.87	-11.60	-0.93
	(23.56)	(17.18)	(22.45)	(5.36)
"Non OHE"	13.76	-11.20	-2.78	0.41
	(23.74)	(21.21)	(23.84)	(3.07)

#### 4.5.2 CALCULUS

## 4.5.2.1 Comparison of the Differences that Occurred in the 4 Groups for Calculus

The mean differences in calculus are shown in table 9. The number of calculus free sites increased in the scaled groups and decreased in the groups that did not receive scaling. There were highly significant differences in calculus score 0 and 2 (F=47.95, p=0.000; F=47.48; p=0.000 respectively) between the groups. There was also a significant difference (F=3.68;p=0.016) between the groups in calculus score 1. There was no significant difference between the four groups in the change that occurred in the number of sites with calculus score 3.

A student Newman-Keul's test indicated that the differences in scores 0 and 2 occurred between the groups that received scaling (groups 1 and 2) and those that did not receive scaling (groups 2 and 3), q=14.16 ; p,0.05. The difference in calculus score 1 occurred between the scaling only group and the other three groups (q=3.59;p<0.05).

#### 4.5.2.2 The effects of scaling on calculus

The number of calculus free sites increased markedly in the "scaled" group while in the "non scaled" group the sites decreased slightly. The number of sites with calculus score 1 (supragingival) and score 3 changed little in both groups.

The mean number of sites with calculus score 2 (subgingival) decreased markedly in the "scaled" group while in the "non scaled" group these sites increased slightly. (table 17) There was a significant difference (F=145.73, p=0.000) between the groups in the change that occurred in the number of sites with calculus score 0. There was no significant difference in between the groups in the change that occurred in the number of sites with calculus score 1 and 3. There was a highly significant difference (F=127.29,p=0.000) between the groups in change that occurred in the number of sites calculus score 2.

Table 17:Mean differences in number of sites with calculusscore 0,1,2,3 in the "scaled" and "non scaled"

g	r	Ō	u	р	s	

		Score		
Treatment Group	0	1	2	3
Scaled (1+2)	37.59	-2.56	-31.69	-3,38
	(19.63)	(4.99)	(19.10)	(9,78)
Non Scaled (3+4)	-3.95	-0.95	5.73	-0.50
	(9.28)	(2.81)	(8.53)	(2.21)

#### 4.5.2.3 The effect of OHE on calculus.

There was little difference in the number of sites with the various scores of calculus between the "OHE" and "non OHE"groups. The number of calculus free sites increased in both groups while the sites with calculus score 1,2,3 decreased in both groups. (table 18).

There were no significant differences between the "OHE"

(group 2+3) and "no OHE" group (group 1+4) in the changes that occurred in the number of sites with any of the calculus scores.

# Table 18:Means of differences in the number of<br/>sites with calculus score 0,1,2,3 in the groups<br/>with OHE (2 and 3) and without OHE (1 and 4)

	Score			
Treatment Group	0	1	2	3
OHE (2+3)	15.50	-0.53	-13,50	-1.20
	(24.40)	(3.25)	(24,37)	(2.67)
Non OHE (1+4)	17.20	-2.49	-12.29	-2.37
	(26.91)	(4.40)	(23.77)	(8.85)

## 4.5.2.4 Comparison of the effects of scaling alone with scaling combined with OHE on calculus

The number of sites with calculus score 0 increased in both the scaling only and scaling combined with OHE groups. The differences in the number of sites with calculus score 0 were similar in both groups. The mean number of sites with calculus score 1,2,3 decreased in both groups. The greatest decrease occurred in the sites with calculus score 2.

There were no significant differences between the groups in the changes that occurred in any of the calculus scores except score 1 (F=3.8, p=0.01). From earlier results of a student Newmwan-Keul's test, the difference in the number of sites with calculus score 1 in the scaling only group was found to be significantly (q=3.59, p<0.05) different from those of the other groups.

The scaling only group had a significantly greater decrease in the number of sites with calculus score 1 than the scaling combined with OHE group.

(1 and 2) and	non scare	a groups		
	Ş	Score		
Treatment Group	0	1	2	3
Scaling only	39.17 (20.56)	-3.92 (5.17)	-30.92 (19.50)	-4.50 (12.30)
Scaling + OHE (2)	35.07 (18.44)	-0.40 (3.96)	-32.93 (19.04)	-1.60 (2.38)
Non scaled (3+4)	-3.95 (9.28)	-0.95 (2.81)	5.73 (8.58)	-0.50 (2.21)

Table 19:Means of differences in the numberof sites with calculus score 0,1,2,3 in the scaled(1 and 2) and non scaled groups

## 4.5.2.5 Comparison of the OHE only group, the scaling combined with OHE group and the "no OHE" group for calculus

The number of calculus free sites (score 0) increased markedly in the group that received scaling combined with OHE while in the OHE only group these sites increased slightly. There was little difference between these two groups in the mean differences in the number of sites with calculus score 1 and 3. The number of sites with calculus score 2 decreased markedly in the scaling combined with OHE group while in the OHE only group these sites increased slightly.(table 20)

#### Table 20: Means of differences in the number of sites with <u>calculus score 0,1,2,3 in groups with OHE (2 and 3)</u> and without OHE (1 and 4).

5	Sec	Score			
Treatment Group	0	1	2	3	
OHE only	-4.07 (8.05)	-0.67 (2.47)	5.93 (7.62)	-0,80 (2,96)	
Scaling + OHE	35.07 (18.44)	-0,40 (3,96)	-32,93 (19,04)	-1.60 (2.38)	
OHE	17.20 (26.91)	-2.49	-12.29 (23.77)	-2.37 (8.85)	

#### 4.5.3 GINGIVITIS

# 4.5.3.1 Comparison of the differences that occurred in the 4 groups for gingivitis

The mean differences for gingivitis are shown in table 10. The number of gingivitis free (score 0) increased in all the groups. This increase was higher in the groups that received scaling. The number of sites with score 1,2,3 decreased in all the groups with the groups that received scaling having the highest increase. There was a significant (F=3.31, p=0.025) difference between the four groups in the change that occurred in the number of sites with gingivitis score 0. There were no significant differences between the groups in the changes that occurred in the number of sites with gingivitis scores 1,2,3.

A student Newman-Keul's test indicated that the difference in score 0 was between the groups that received scaling and those that did not receive scaling (3 and 4), (q=3.69; p<0.05).

#### 4.5.3.2 The effect of scaling on gingivitis

The number of gingivitis free sites (score 0) increased in both the "scaled" and "non scaled" groups. The "scaled" group had a higher increase. The number of sites with gingivitis score 1,2,3 decreased in both groups. The decrease was higher in the scaled group. (table 21)

The changes in gingivitis score 0,1,2 in the "scaled" and "non scaled" groups were significantly different ( F=9.23, p=0.003; F=6.33, p=0.014; F=5.32, p=0.024) respectively.

Table 21:	<u>Means of the differences in the number</u>
	of sites with gingivitis score 0,1,2,3 in the
	scaled (1 and 2) and non scaled (3 and 4) groups.

	Score			
Treatment Group	0	1	2	3
Scaled (1+2)	21.64	-16.67	-4.85	-0.08
	(24.71)	(23.12)	(8.63)	(0.35)
Non scaled (3+4)	6.95	-5.10	-1.53	0.0
	(17.79)	(17.39)	(2.87)	(0.25)

#### 4.5.3.3 The Effect of OHE on Gingivitis

The number of gingivitis free sites (score 0) increased in both the "OHE" and "non OHE" groups. The number of sites with gingivitis score 1,2,3 decreased in both groups. The mean differences were not very different between the groups although they tended to be slightly higher in the "OHE" group.(table 22) There were no significant differences between "OHE" group and "no OHE" group in the changes that occurred in the number of sites with of any of the gingivitis scores.

Table 22:	Means of the differences in the number	
	of sites with gingivitis score 0,1,2,3 in the	
	groups with OHE (2 and 3) and without OHE(1 and 4)	1

	Score			
Treatment Group	0	1	2	3
OHE (2+3)	16.87	-12.10	-4,50	-0.07
	(22.10)	(20.99)	(4,87)	(0.37)
Non OHE (1+4)	12.57	-10.02	-2.35	-0.02
	(22.96)	(21.36)	(7.35)	(0.25)

## 4.5.3.4 Comparison of the Effects of Scaling alone with Scaling Combined with OHE on Gingivitis

The number of sites with gingivitis score 0 increased in both the scaling only and scaling combined with OHE groups. The number of sites with gingivitis score 1,2,3 decreased in both groups. The mean differences tended to be higher in the scaling combined with OHE group. (table 23).

There were no significant differences between the groups in the changes that occurred in the number of sites with ginglvitis score 0,1,2,3.

	Score			
Treatment Group	0	1	2	3
Scaling only (1)	19.29	-15.08	-4.13	-0.04
	(23.62)	(22.04)	(9.98)	(0.20)
Scaling + OHE (2)	25.40	-19.20	-6.00	-0.13
	(26.78)	(25.04)	(6.00)	(0.52)
Non Scaled (3+4)	6.95	-5.10	-1.53	0.0
	(17.79)	(17.39)	(2.87)	(0.23)

Table	23:	Means	of t	he (	differer	ices	in	gingivi	tis	SCOL	ce
		0,1	,2,3	in	groups	1,	2	(scaled)	and	13,	4
		(non	i sca	led	2						

## 4.5.3.5 Comparison of the group that received OHE only, scaling combined with OHE and the "no OHE" group for gingivitis

The mean differences in number of sites with gingivitis score 0 increased in both the OHE only and the scaling combined with OHE groups by a similar number of sites. The mean differences in the number of sites with gingivitis score 1,2,3 decreased in both groups. The scaling combined with OHE group tended to have a higher decrease. (table 24)

There were no significant differences between the groups in the changes that occurred in the number of sites with score 0,1,2,3.

Table 24	: <u>A comparison of means of differences</u>
1. Con	in number of sites with gingivitis score
	0,1,2,3 between groups 1, 2 (scaled) and
	3,4 (non scaled).

	Score				
Treatment Group	0	1	2	3	
OHE only (3)	25.40	-5.00	-3,00	-0.13	
	(8.33)	(25.34)	(2,88)	(0.52)	
Scaling + OHE (2)	26.78	-19.20	-6.00	0.0	
	(11.79)	(12.75)	(6.00)	(0.0)	
Non Scaled (3+4)	12.57	-10.02	-2.35	-0.02	
	(22.96)	(21.36)	(7.35)	(0.25)	

#### 4.5.4 POCKET DEPTH

#### 4.5.4.1 Comparison of the differences that occurred in the 4

#### groups for pocket depth

The mean differences for pocket depth are shown in table 11. The mean differences that occurred in the number of sites with shallow and deep pockets in the groups that received scaling were few (2-3 sites). The shallow pockets increased and the deep pockets decreased. In the groups that did not receive scaling there was hardly any change in the pockets.

Although an analysis of the mean differences indicated that there was a significant difference (F=2.87; p=0.042) between the groups in the change that occurred in the deep pockets, a student Newman-Keul's test indicated that there were no significant differences.

#### 4.5.4.2 The effect of scaling on pocket depth

The mean differences in the number of sites with shallow and deep pockets changed by a few sites (2) in the "scaled" group. The shallow pockets increased and the deep pockets decreased. There was hardly any change in the pockets in the "non scaled" group. (table 25).

The "scaled" group had a significantly (F=6.37, p=0.014) greater increase in the number of shallow pockets than the "non scaled" group. The "scaled" group also had a significantly (F=8.57, p=0.005) greater decrease in the number sites with deep pockets.

Table 25:Means of differences in number of<br/>sites with shallow (≤3 mm) and deep (≥4 mm)<br/>pockets in the scaled (1 and 2) and non scaled<br/>(3 and 4) groups.

	Score		
Treatment Group	≤3 mm	≥ 4 mm	
Scaled	2.41 (3.38)	-2.33 (3.19)	
Non scaled	0.63 (2.90)	-0.38 (2.74)	

#### 4.5.4.3 The effects of OHE on the pocket depth

The mean difference in the number of sites with shallow and deep *pockets was small (1-2 sites) in both the "OHE" and "non* OHE" groups. The number of shallow pocket increased while the deep pockets decreased in both groups by a similar number of sites. (table 26).

There were no significant differences between the "OHE" group (group 2+3) and the "no OHE" group in the changes that occurred in the shallow or deep pockets.

# Table 26:Means of differences in number<br/>of sites with shallow (≤3 mm) and deep (≥4 mm)<br/>pockets in the "OHE" and "no OHE group

	Score		
Treatment Group	≤3 mm	≥4 mm	
"OHE" (2+3)	1.53 (3.16)	-1.43 (3.14)	
"Non OHE" (1+4)	1.49 (3.34)	-1.29 (3.13)	

## 4.5.4.4 Comparison of the effect of scaling alone with scaling

#### combined with OHE on the pocket depth

The increase in the number of sites

with shallow pockets was similar in the scaling only and the scaling combined with OHE groups. The decrease in the number of sites with deep pockets was also similar in both groups. (table 27)

There were no significant differences between the scaling only and the scaling combined with OHE groups in the changes that occurred in the shallow and deep pockets.

# Table 27:Means of differences in number of sites with shallow $(\leq 3 \text{ mm})$ and deep $(\geq 4 \text{ mm})$ pockets in "scaled"<br/>groups (1,2) and the "non scaled" group(3,4).

	Score		
Treatment Group	≤3 mm	≥4 mm	
OHE only (1)	2.29 (3.43)	-2.17 (3.06)	
OHE + scaling (2)	2,60 (3,40)	-2,60 (3,48)	
Non OHE	0.63 (2.90)	-0.38 (2.74)	

## 4.5.4.5 Comparison of the OHE only group, the scaling combined with OHE group and the "no OHE" group for pocket depth

The OHE only group had little change in the pocket depth. In the scaling combined with OHE and the "non OHE" groups, the number of sites with shallow pockets increased and the deep pockets decrease by 1-2 sites. (table 28)

There were no significant differences between the three groups in the changes that occurred in the pocket depth. Table 28: Means of differences in number of sites with shallow ( $\leq 3$  mm) and deep ( $\geq 4$  mm) pockets in groups with (2,3) and without OHE (1,4).

he	Score			
Treatment Group	≤3 mm	≥4 mm		
OHE only (1)	0.47 (2.59)	-0.27 (2.31)		
Scaling + OHE (2)	2.60 (3.40)	-2.60 (2.31)		
Non OHE (3+4)	1.49 (3.34)	-1.29 (3.13)		

#### 4.5.5 ATTACHMENT LEVEL

# 4.5.5.1 Comparison of the differences that occurred in the 4 groups in the attachment level.

The mean differences for attachment level are shown in table 12. The number of sites with attachment loss 0 mm increased in all the groups while the number of sites with attachment loss 1-3 mm decreased by a similar number of sites. The number of sites with attachment loss 24 mm decreased by a few sites in the groups that received scaling while in the groups that did not receive scaling hardly any change occurred.

There were no significant differences between the groups in the changes that occurred in the number of sites with attachment scores 0 mm, 1-3 mm,  $\geq 4$  mm.

#### 4.5.5.2 The effects of scaling on the attachment level

The number of sites with attachment loss 0 mm increased in both the "scaled" and "non scaled" groups. The increase was higher in the "scaled" group. The number of sites with attachment loss 1-3 mm decreased in both groups with the "scaled group having the greater decrease. The number of sites with attachment loss ≥4 mm decreased slightly in the "scaled" group while in the "non scaled" group there was very little change. (table 29) The changes that occurred in the "scaled" and "non scaled group in the number of sites with attachment loss 0 mm were significantly (F=6.98, p=0.010) different. The changes that occurred in both groups in the number of sites with attachment loss 1-3 mm and ≥4 mm were also significantly different (F=4.60,p=0.035; F=5.60, p=0.021 respectively.

		Score	
Treatment Group	0 mm	1-3 mm	≥4 mm
Scaled (1+2)	38.08 (18.29)	-36.54	-1,44
Non scaled (3+4)	28.45 (13.85)	-28,48 (14,85)	0.35

Table 29: Means of differences in number

of sites with attachment level 0 mm, 1-3 mm and

24 mm in the "scaled" and "non scaled" groups.

68

#### 4.5.5.3 The effects of OHE on the attachment level

The changes that occurred in the attachment level were similar in the "OHE" and "non OHE" groups. The number of sites with attachment loss 0 mm increased in both groups while the number of sites with attachment loss 1-3 mm decreased by a similar number of sites. The number of sites with attachment loss 24 mm decreased by a few sites in both groups.(table 30) There were no significant differences between the "OHE" group (group 2+3) and the "no OHE" group (group 1+4) in the changes that occurred in the number of sites with attachment loss 0 mm, 1-3 mm, and 24 mm.

Table 30: Means of differences in number of sites with attachment level 0 mm, 1-3 mm and >4 mm in the "OHE" and "no OHE" groups.

	Score		
Treatment Group	0 mm	1-3 mm	.≥4 ,mm
"OHE" (2+3)	35.53	-34.70	-0.67
	(18.12)	(18.69)	(3.16)
"Non OHE" (1+4)	31.76	-31.08	-0.45
	(15.97)	(16.08)	(3.65)

## 4.5.5.4 Comparison of the effect of scaling alone with that of scaling combined with OHE on the attachment level

The changes in the number of sites having attachment level score 0 mm, 1-3 mm, and 24 mm was similar in the "scaled" and "non scaled" groups. The number of sites with attachment loss 0 mm increased in both groups while the number of sites with

attachment loss 1-3 mm decreased by a similar number. The number of sites with attachment loss  $\ge 4$  mm decreased by a few sites in both groups. (table 31)

There were no significant differences between the groups in the changes that occurred in attachment level 0 mm, 1-3 mm,  $\ge 4$  mm.

# Table 31:Means of differences in number of<br/>sites with attachment levels 0 mm, 1-3 mm and $\geq 4$ mm<br/>in the "scaled" (1,2) and the "non scaled" (3,4)<br/>groups

	Score			
Treatment Group	0 mm	1-3 mm	.≥4 mm	
Scaling only (1)	37.08	-35,67	-1.25	
	(18.58)	(18,52)	(3.37)	
Scaling + OHE (2)	39.67	-37.93	-1.73	
	(18.36)	(18.85)	(2.58)	
Non Scaled (3+4)	28.45	-28.48	0.35	
	(13.84)	(14.84)	(3.62)	

4.5.5.5 Comparison of the OHE only group (group 3), scaling combined with OHE group (group 2) and the "no OHE" group (group 1+4)for the attachment level

The number of sites with attachment loss 0 mm increased in all three groups. The scaling combined with OHE group had a slightly higher increase than the other two groups. The number of sites with attachment loss 1-3 mm decreased in all the groups by a similar number of sites as those that increased in attachment loss 0 mm. The decreased was slightly higher in the scaling combined with OHE group than in the other two groups. The number of sites with attachment 24 mm hardly changed in the OHE only and the "non OHE" groups while in the scaling combined with OHE group, the sites decreased by about 2 sites (table 32).There were no significant differences between the groups in the changes that occurred in the number of sites with attachment loss 0 mm, 1-3 mm, 24 mm.

# Table 32:Means of differences in number of sites with attachment<br/>levels 0 mm, 1-3 mm and 24 mm in the OHE, scaling<br/>and without OHE (1,4).

Treatment Group	Score			
	0 mm	1-3 mm	≥4 mm	
OHE only (3)	31.40 (17.51)	-31.46(18.60)	0.40 (3.40)	
Scaling + OHE (2)	39.67 (18.36)	-37.93 (18.85)	-1.73 (2.58)	
Non OHE (1+4)	31.78 (15.97)	-31.08 (16.08)	0.45 (3.65)	

#### CHAPTER 5

#### 5. DISCUSSION CONCLUSION, RECOMMENDATIONS AND LIMITATIONS OF

#### THE STUDY

This chapter contains the discussion of the results, the conclusion and the recommendations of the study. Some of the limitations of this study have also been outlined in this chapter.

#### 5.1 DISCUSSION

#### 5.1.1 General Characteristics:

The age range of this study population was within the range where most signs of periodontal are expected (Pilot.T.et al,1986). Although majority of the participants in this study claimed to brush their teeth regularly, the oral hygiene was found to be quite poor with an average of only 24 plaque free sites per individual. It was felt that subjects answered the question about the frequency of tooth brushing according to how often they thought they should have been brushing rather than what was actually done.

The total number of mobile teeth in this study group was 36 i.e. roughly one mobile tooth in every third or fourth individual examined. However, all the mobile teeth were observed to have occurred in a few individuals only. The total number of missing teeth was 123. Most (63%) of these had been lost due to caries rather than periodontal disease. Periodontal disease contributed to 16% of all the lost teeth. Other recent studies have similarly found caries to be a greater cause of tooth loss than periodontal disease (Ainamo, J et al, 1984; Manji, F et al, 1988). Traditional extraction was found mainly in the older individuals (over 40 year olds) and contributed to 24% of all the missing teeth. Traditional extraction did not occur in those less than 35 years.

#### 5.1.2 Distribution of the participants in the treatment groups

After randomization, the individuals were fairly evenly distributed among the four groups (table 1). Data on subjects who had been initially randomized to receive scaling, OHE or a combination of both, and who did not avail themselves for the interventions, were analyzed together with "no treatment" group (group 4). The other options would have been either analyzing their results with those of the groups to which they had been randomized or excluding them from the analysis completely. The combination with the "no treatment" group was felt justifiable since the baseline (pre-treatment) characteristics of the subgroup did not significantly differ from those of the primary groups of randomization. Analysis with initial groups of randomization would not have been clinically meaningful since they had no therapy at all while, on the other hand, total exclusion would have reduced the total sample size and hence the power of the study.

A total of 22 subjects dropped out of the study before the final evaluation. Reasons for dropping out included fear of being scaled and failure to attend the final examination due to

disappointment on realising that they had been randomized to the "no treatment" group. The latter may explain the high drop out rate in the "no treatment" group. Blinding was, unfortunately, not practicable in this kind of a study and subjects were therefore, aware of their groups of randomization.

#### 5.1.3 Comparison of the groups at baseline

There were no significant differences between the groups in any of the periodontal parameters. In the groups majority of the sites were found to have plaque. Most of these sites had small amounts of plaque (score 1). There were only a few sites with great amounts of plaque (table 2). This result was thought to have occurred because many of the individuals attempted to clean their teeth before the examinations despite efforts to discourage them from doing so. Had this not happened, probably sites with plaque score 2 and 3 would have been found in these individuals.

Most of the participants in the four groups had calculus on more than half the number of sites in their mouth (table 3). The amount of calculus present in an individual was found to increase with age with those under 35 years having little calculus and those over 40 years having a lot more calculus. Most of the sites had subgingival calculus. The number of sites with supragingival calculus were relatively few and occurred mainly on the lower anterior teeth. Both subgingival and supragingival calculus occurred on the same surface on only a few sites. These were mostly interproximal surfaces of the low anterior teeth.

In majority of the participants in the four groups, gingival bleeding occurred on more than half their sites (table 4). Most of the bleeding was mild. There were very few sites having pus in this study population.

The majority of the pockets in this study group were shallow. The mean number of shallow pockets in all the four groups was over 100 (table 5). This resulted from one of the exclusion criteria that any individual with more than four teeth having a site with a pocket 24 mm was excluded from the study. However, it was observed that these individuals were quite few in the study population. In each of the factories, there were 3-5 such subjects in the study population. The attachment loss was mainly 1-3 mm. Again this may have occurred due to the exclusion criteria mentioned above. Most of the attachment loss in this study population was observed to occur on the lingual aspects of the teeth. In many sites with attachment loss greater than 4mm gingival recession had occurred and the associated pockets were not deep.

# 5.1.4 Differences in periodontal parameters within the four groups between the baseline and final examination.

#### Plaque

There was a significant increase in the number of plaque free sites in all the four groups at the final examination (table 8). Although the groups that received OHE (groups 2 and 3) were expected to have a greater increase in the number of plaque free sites than the groups that did not (groups 1 and 2), there were no significant differences between these groups in the increased number of plaque free sites. This may have resulted due to a filtering of the OHE from the group that received it to those that did not. It was quite possible for this to occur because the individuals of these groups work together. The increase in the plaque free sites in the groups that did not receive OHE may also have occurred because of an increase in awareness of their oral hygiene due to their involvement in the study. This awareness may have been encouraged by the fact that all the participants in the study were given free tooth brushes and toothpaste as an incentive for participating in the study.

#### Calculus

The number of calculus free sites (score 0) increased significantly in the groups that received scaling (groups 1 and 2) while in the groups that did not receive scaling (group 3 and 4) these sites decreased slightly (table 9). This decrease was mainly due to a decrease in the number of sites with subgingival

calculus. There was no significant change in the number of sites with supragingival calculus in the scaling combined with OHE group. However, surprisingly there was a significant decrease in these sites in the scaling only group. The reason for the scaling only group having a significantly greater decrease in the number of sites with supragingival calculus than the scaling combined with OHE group are not clear but it is an indication that the OHE was not effective. Had the OHE been effective we would have expected the plaque control in the scaling combined with OHE group to have been better than in the scaling only group and the reformation of the supragingival calculus would therefore have been lower in the former group.

In the groups that did not receive scaling the number of sites with calculus increased significantly. This increase was mainly due to a significant increase in the number of sites with subgingival calculus in these groups. The number of sites with supragingival calculus did not change significantly in these groups.

#### Gingivitis

A significant increase in the gingivitis free sites (score )) occurred in the groups that received scaling. There was no significant change in the groups that did not receive scaling. Scaling therefore appears to have been effective in reducing the number of sites with gingival bleeding probably by creating a change in the subgingival flora in these sites. It is surprising

that this change could have been maintained for six months since the general OHE in both the groups that received scaling was poor. This may be an indication that gingival bleeding has a greater association with calculus than with supragingival plaque.

#### Pocket depth

A significant change in pocket depth occurred in only a few sites in the scaled groups. This was mainly because there were few sites with pockets 24 mm. Studies in literature have indicated that most of the change in pocket depth after scaling occurs in deep pockets (Lindhe.J.et al 1987). Sterne, J., in a paper published in the proceedings of a conference on periodontal disease in 1988, showed mathematically that a greater change can be expected in deeper pockets because of a phenomenon of regression towards the mean. There was no significant change in pocket depth in the groups that did not receive scaling.

#### Attachment level

The mean number of sites with attachment loss 0 mm increased by the a similar number of sites as those that decreased in the sites with attachment loss 1-3 mm. This was likely to have occurred due to a change in the measurement criteria between the baseline and final examination where an attachment level which was read 1-3 mm at baseline was read 0 mm in the final examination even though it had remained unchanged. This was quite possible because the measurement error for the attachment level

in this study was found to be  $\pm 3$  mm. It was therefore difficult to differentiate between the 0 mm and 1-3 mm pockets.

# 5.1.5 The effects of scaling and OHE on the periodontal parameters

#### Plaque

Scaling did not have any significant effect on plaque. There was no significant difference between the scaled and "non scaled" group in the change that occurred in the plaque free sites (table 13). We would not have expected scaling to have produced much effect on plaque since it was done only once during the six months of the study. Improvement on the plaque free sites could only have been expected if the oral hygiene of the scaled individuals was good. However, scaling appears to have caused a significant decrease in the number of sites with plaque score 2 suggesting that it may have had an effect in reducing the amount of plaque on the tooth surfaces. This may have occurred because scaling reduced the number of sites with calculus which acts as a retention for plaque.

The OHE in this study did not create a significant reduction in the plaque free sites. There was no significant difference found between the "OHE" and "non OHE" group (table 14). The number of plaque free sites increased in both groups. As discussed earlier, this reduction may have been due to filtering of the OHE from the groups that received it to those that did not or it may have occur due an increase in awareness in oral hygiene

in all the groups resulting from their involvement in the study. Whichever was the case, the OHE was not very effective because despite a significant increase occurring in the plaque free sites by about 21 sites the mean number of plague free sites after treatment was still less than half the total number of sites (assuming that the total number of sites per individual to be about 112 since the average tooth loss in the study population was less than 1 i.e 0.37) since the mean number of plaque free sites at in the groups at baseline was 22-26. For the OHE to have been effective a behavioral change in the participants was necessary. Studies in health education have found that for a behavioral change to occur in an individual, it is necessary to recall him a number of times for feedback and reinforcement (Woodhall, I.R, 1984). The OHE in this study was given only once and was therefore not adequate to produce a behavioral change in the participants. In this study, the OHE was only given once because it was felt that to give it more frequently would have unpractical on a public health basis.

A combination of scaling and OHE did not have any advantage over scaling alone in reducing plaque in this study. There was no significant difference between the group that received scaling only and that which received scaling in addition to OHE in the changes that occurred in plaque (table 15).

#### Calculus

Scaling resulted in a significant increase in the number of calculus free sites (table 17). This was mainly due to an decrease in the sites with subgingival calculus. The effect of scaling on supragingival calculus was not maintained over a six months period. In fact in the study, supragingival calculus was found to have reformed in some individuals at the time the OHE was been given i.e two weeks after scaling. The number of sites with supragingival calculus had returned to baseline levels at the final examination in the scaling combined with OHE group. OHE did not prevent the formation of calculus as may have been expected. The number of sites with supragingival calculus returned to baseline levels at the final examination in the scaling combined with OHE group. In the OHE only group the number of sites with subgingival increased significantly at the final examination (table 20). This is another indication that the OHE in this study was not very effective in creating an improvement in the plague control in the participants.

#### Gingivitis

Scaling significantly reduced the number of sites with gingival bleeding. The number of gingivitis free sites were significantly fewer in the "scaled" than in the "non scaled" group (table 21). It was surprising that a significant reduction in gingivitis was maintained for six months considering that the plaque control in the scaled groups was poor. In their study on

experimental gingivitis, Loe.H et al,1965, found that gingivitis occurs within a few days where plaque control was poor. The above results may suggest that gingival bleeding has a greater association with calculus than with plaque. However, Mandel I.D et al,1986, in their review quoted studies which indicate gingival bleeding has a greater association with plaque than with calculus. Another possibility is that the gingival bleeding may depend on the amount of plaque on a surface. Low quantities of plaque (score 1) on a surface may not cause gingival bleeding while great quantities of plaque (score 2) result in gingival bleeding. This would explain the result that although the number of plaque free sites was not significantly different between the "scaled" and "non scaled"groups, gingival bleeding was reduced significantly in the "scaled" group where the number of sites with plaque score 2 also decreased significantly.

OHE did not have a significant effect on gingivitis in this study. There were no significant differences in the gingivitis scores between the "OHE" and "non OHE" groups(table 22). Combining scaling with OHE did not have any advantage over scaling alone for gingivitis. There were no significant differences between both groups for any of the gingivitis scores (table 23).

#### Pocket depth

Scaling caused a statistically significant improvement in pocket depth although the number of pockets involved were few.

This limited change in the pockets may have been due to the fact that majority of the pockets in the study groups were few.

OHE did not have any effect on the pocket depth and there are no significant differences between the "OHE" and "non OHE" groups in the changes that occurred in the pockets. Combining scaling with OHE did not have any advantage over scaling alone for pocket depth. No statistically significant differences occurred between the two groups (table 27).

#### Attachment

Scaling created a significant reduction in the attachment level 24 mm. The greatest improvement in pockets after scaling has been found to occur in the deep pockets (Philsrom et al, 1984). As stated above these pockets were few in these groups. OHE did not have a significant effect on the attachment level and no differences were observed between the "OHE" and "non OHE" groups. Combining scaling with OHE did not have any significant over scaling alone for the attachment level. No significant differences occurred between the two groups (table 31).

#### 5.2 CONCLUSION

Scaling in this study produced some improvement on all the periodontal parameters except plaque. Its effect on the pocket depth and attachment level was limited in this study population. The main effect of scaling was on the subgingival calculus but

the supragingival calculus did not improve. The number of gingivitis free sites improved significantly after scaling. The OHE in this study was not effective in creating an improvement in plaque control in the relevant groups. Therefore, the groups that received OHE did not differ significantly from those that did not in any of the parameters.

Combining scaling with OHE did not have any advantage over scaling alone in this study. This was due to the failure in the OHE. The group that received scaling combined with OHE did not differ significantly from the scaling only group in any of the parameters.

Although in this study scaling was found to be effective in improving the periodontal status, it doubtful if this improvement would have been maintained over a long period of time since the levels of plague in the scaled subjects did not improve. Indeed, supragingival calculus was found to have reformed within six months. It is possible that a deterioration also occurred in the other periodontal parameters, however, this was not detectable since there was only one post-treatment examination.

#### 5.3 RECOMMENDATIONS

A study with more post-treatment examinations is required so that the changes that occur with time in the periodontal parameters after scaling can be determined. A deterioration on the supragingival calculus was observed to have occurred within the six months of this study. Some deterioration of the other

parameters was suspected. The deterioration was attributed to the poor plaque control in the subjects that recieved scaling. A study to determine the best methods of improving plaque control on a community basis is required. So far, no such study has been done in Kenya.

A longer study is required in order to determine whether scaling had any effect on the progression of periodontal disease. The six months of this study were too short for a significant change in the attachment level to occur.

It was interesting to observe that most of individuals in this study population had retained most of their teeth in a functional state despite the presence of untreated periodontal disease. Manji et al (1988), in their study on a Kenyan population found that despite having had no preventive periodontal treatment, majority of the population retained most of their teeth in a functional state over the age of sixty five years. If this is the case, then the benefits of scaling at the community level in Kenya, a country with an estimated life expectancy of 65 years at birth, are not be very clear. More studies on the progression of periodontal disease in Kenyans are therefore, required in order to determine treatment requirements for Kenya.

More practical objectives for periodontal therapy on a public health basis should be sort. The current objective of eliminating all signs of disease (WHO, 1986) is not practical and may not be necessary.

#### LIMITATIONS OF THE STUDY:

1.

2

In this study the author did the treatments and examinations. This was because of a problem in obtaining a technically competent person to do these procedures. The suitable personnel that could have been used would either have been dental students or dentists working in the public sector. It was not possible to use dental students because the involvement and the amount of time required for the study was not compatible with their school curriculum. Employing a dentist would have been too expensive for the study.

However, the feeling when this study was conducted was that the length of time between the treatments and the final examination (6 months) and recall of the group in which an individual belonged would have been difficult at the final examination.

Another limitation of this study is that it was designed such that those that received OHE and those that did not receive OHE worked in the same work place. This meant that there was a possibility of OHE filtering to the groups that were not supposed to receive it (contamination). However, it was felt that separating these groups by using a different factory for each group may have made the groups incomparable in certain important baseline characteristics.
- Having only one examination after treatment also created a 3. limitation in the study. This is because the progression of effects of the treatment could not be determined. This study was initially designed to have two examinations after treatment, one three and another six months after. However, some of the workers were temporary laid off work due to a crisis in wheat shortages in the country around the time the 3 month examination was to be carried out. It was therefore not carried out and it was not possible to plan for another examination because of a limitation in the time available to carry out the research. This meant that it was not possible to tell from the study if the effects of scaling remained at the final examination as immediately after the same treatment or whether they deteriorated and if they deteriorated it was not possible to tell by how much this occurred.
  - Another limitation to the study related to the provision of OHE on a single occasion. While it was not possible to offer *more than one session in the present study due to financial* limitations, more than one session might be necessary to ensure adequate comprehension of instruction. Furthermore, long term control of periodontal disease requires sustained change in oral hygiene and therefore need for a more intensive health education programme.

#### REFERENCES

- Akpabio, S.P. (1970): Dentistry a Public Health Service in East and West Africa. <u>M.D.S. Thesis, University</u> of London.
- Ainamo, J, Sarkki, Kuhalampi, M.L., Palolampi L., and Piirto O. (1984): The frequency of periodontal extractions in Finland. <u>Community Dental Health:</u> 1; 165-172.
- Axelsson, P., Lindhe, J., (1978): Effect of controlled oral hygiene procedures on caries and periodontal disease in adults. <u>Journal of clinical periodontology</u>, 5:133-151.
- Badersten, A.' Nilveus, R.' Egelberg, J.' (1981): Effect of non-surgical periodontal therapy I. Moderately advanced periodontitis. <u>Journal of clinical periodontology</u>. 8:57-72.
- Badersten, A., Nelveus, R., and Egelberg, J. (1984): Effects of non surgical periodontal therapy II. Severely advanced periodontitis. <u>Journal of clinical periodontology</u>. 11:63-67.
- Baelum, V., Fejerskov, O., and Karring, T., (1986): Oral hygiene, Gingivitis and Periodontal Breakdown in adult Tanzanians. <u>Journal of periodontal research. 21:221-232</u>.
   Baelum, V., Fejerskov, O., and Manji F. (1988): Periodontal diseases in adult Kenyans. <u>Journal of</u>

clinical periodontology. 15:445-452.

- Ballit, H., Brown R., (1987): Is periodontal disease the primary cause of tooth extraction in adults? <u>J.A.D.A.,</u> Vol 114 pp 40-45.
- Barmes, D.E., (1986). Epidemiology of Periodontal Diseases. <u>Acta Paradontologica in Schweiz, Maschr</u> Zahmed 96:908/88-911/91.
- 9. Becker, W., Becker, E.B., and Berg, L.E., (1984): Periodontal Treatment without plaintenance. A retrospective study in 44 patients. <u>Journal of</u> <u>Periodontology 55(9): 505-511.</u>
- 10. Cerek, J.F., Kiger, R.D., Garrett, S., and Egelberg J. 1983). Relative effects of plaque control and instrumentation on the clinical parameters of human periodontal disease. <u>Journal of clinical periodontology</u> 10:46-56.
- 11. Eide, B., Lie T., and Selvig, K.A., (1983): Surface coating on dental cementum incident to periodontal disease. <u>Journal of clinical periodontology 10:157-171.</u>
- 12. Goldman, H., (1986): Periodontal Disease, Part IV: Calculus and other aetiologic factors. <u>The compendium of</u> <u>continuing Education 7 (4): 270-278.</u>
- 13. Goodson, J.M. Yanner, A.C.R., Haffajee, A.D., Sornbergen G.C., and Socranssky, S.S. (1982): Patterns of progressions and Regression of advanced destructive periodontal disease. <u>Journal of Clinical Periodontology</u> 9:472-481.

- 14. Jackson, A.P., and Laws A.J. (1986): Toothloss amongst 70 treated periodontal patients. <u>Journal of the New</u> Zealand Society of Periodontology 61:5-7.
- 15. Kaimenyi, J.T., Sachdeva, P., Patel S., (1988): Cause of tooth mortality at the Dental Unit of Kenyatta National Hospital of Nairobi, Kenya. <u>Tropical Dental Journal</u> 1:17-20.
- 16. Lindhe J., Nyman, S., and Karring, T., (1982): Scaling and root planning in shallow pockets. <u>Journal of</u> <u>clinical periodontology 9:415-418.</u>
- 17. Lindhe. J., Westfelt, E., Nyman, S., Socransky, S.S., Heijl, L., and Bralthall, G., (1982): Healing following surgical/non-surgical treatment of periodontal disease. <u>Journal of Periodontology 9:115-128</u>.
- 18. Lindhe, J., Haffajee, A.D., and Socransky, S.S., (1983): Progression of periodontal disease in adult subjects in the absence of periodontal therapy. <u>Journal of clinical</u> periodontology 10:433-443.
- 19. Lindhe, J., Socransky, S.S., Nyman, S., and Westfelt E., (1987): Dimensional alternation of the periodontal tissues following therapy. <u>The International Journal of</u> <u>Periodontics and Restorative Dentistry 2:9-19.</u>
- 20. Loe, H., Theilade, E., and Jensen S.B. (1965): Experimental gingivitis in Man. Journal of Periodontology 36:177-187.

- 21. Loe, H., Anerud, A., Boysen, H., and Smith M., (1978): The natural history of periodontal disease in man. The rate of periodontal destruction before 40 years of age. Journal of Clinical Periodontology 13:431-440.
- 22. Loos, B., Claffey, N., and Crigger M. (1988). Effects of oral hygiene measures on clinical and microbiological parameters of periodontal disease. <u>Journal of Clinical</u> Periodontology 15:211-216.
- 23. Mandel, I.D., and Gaffar, A., (1986): Calculus Revisited - a review. <u>Journal of Clinical Periodontology</u> <u>13:249-257.</u>
- 24. Manji, F., and Sheiham, A., (1986): CPITN findings and manpower implications of periodontal treatment needs for Kenyan children. <u>Community Dental Health 3:143-151</u>.
- 25. Manji, F., Baelum, V., Fejerskov, O., (1988): Tooth mortality in an adult rural population in Kenya. <u>Journal of</u> <u>Dental Research, 67(2): 496-500.</u>
- 26. Ostein, B., Henrikson, C.O., Nordenram, A.m (1985): Periodontal Health of an Isolated tribe in the Cameroons. Odontostomatol Trop. VIII (1).
- 27. Olsson B., (1978): Periodontal disease and oral hygiene in Arussi Province, Ethiopia, <u>Community Dent. Oral Epidemiology</u> <u>6:139-145.</u>
- 28. Owino, G.R. (1984): The role of the Dental profession in meeting Kenyan's Dental Problems, <u>Kenya Dental Association</u> <u>Newsletter (Jan-Mar) pp 8-11.</u>

APPE	NDIX I
	History Sheet
1.	Name:
2,	Age:
3.	Tribe:
4.	Residential Area:
5. trea	Do you have any medical problem for which you are being ted e.g. a) diabetes b) epilepsy c) any other (specify)
6.	Do you brush your teeth? Y/N
7.	If you do, a) how many times a week? b) how many times per day?
8.	What type of brush do you use? a) Manufactured toothbrush b) Mswaki c) Any other (specify)
9.	What class did you reach in school?
10.	Do you smoke cigarettes? Y/N
11.	If you do, how many do you smoke per day?
12.	Do you drink alcohol? Y/N
13.	If you do:a) what type (beer, whisky, changaa) b) how many times a week? c) how much at one sitting?
14.	Do you chew miraa? Y/N
15.	If you do, a) how often? b) how much? (bundles)
16.	Which toothpaste do you use?
17.	Have you had any dental treatment? Y/N

# APPENDIX 11:

# ORAL HEALTH RESEARCH UNIT Data Collection Form

Study ref:			1					I	III	H	IH/N	0				
Case No:																
Disco (Schoo	1		[								U	rba	n	1		
Place/Schoo	1										R	lura	1	2		
Evaminer:			$\frac{1}{2}$									Ma1	e	1	[	
Draminer.			3									Fem	ale	2	L	
												Age	:			
Teeth Statu	s:															
		a 4														
1	8	7	6	5	4	3	2	1	1	2	3	4	5	6	7	8
Dental Fluo	rosi	s:														
				1												
8	7	6	5	4	3	2	1	1	2	3	4	5	6	7	8	
																Fci





Case No:	1		Date:		
Plaqu M F D	e L	Calculus M F D L	Gingiva M F D L	Attachment M F D L	Pocket M F D L
18       17       16       15       14       13       12       11       21       23       24       25       26       27       28	$ \begin{array}{c} 18\\17\\16\\15\\14\\13\\12\\11\\22\\23\\24\\25\\26\\27\\28\end{array} $		18         17         16         15         14         13         12         11         21         23         23         24         25         26         27         28	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	
M F D	L 48 47 46 45 44 43 42 41 31 32 33 34 35 36 37 38		M F D L 48 47 46 45 44 43 42 41 31 32 33 34 35 36 37 38	M F D L 48 43 47 46 46 45 45 44 43 42 42 41 31 31 32 33 34 34 35 36 36 37 38 38	M F D L
Mobility	18 17 1 48 47 4	6 15 14 1	3     12     11     21     2       1     1     1     1     1       3     42     51     31     3	2       23       24       25       26       2         2       33       34       35       36       3	7 28

#### APPENDIX III: Guideline for Oral Hygiene Instructions

- A. Causes of periodontal disease and caries
  - Poor oral hygiene resulting in plaque accumulation.
     This plaque contains bacteria which cause problems.
  - Frequent ingestion of sugary foods. This sugar together with bacteria from plaque lead to formation of caries.
- B. Poor Oral Hygiene may be due to:
  - 1. Not brushing at all
  - 2. Infrequent brushing
  - 3. Ineffective brushing.
- C. Signs of dental diseases
  - pain on eating or drinking
  - bleeding gums
  - swollen gums
  - gum recession
  - pus exudation from gums
  - smelly mouth
  - tooth mobility
- D. To prevent dental diseases the following procedures should be followed:-
  - 1. Brush teeth daily with a good toothbrush or mswaki,

- 2. Have a systematic way of brushing to ensure all teeth surfaces have been brushed e.g. brushing facial surfaces of the upper teeth followed by the lingual surfaces and the same procedure in the lower teeth, and finally occlusal surfaces up then down.
- Use short horizontal strokes to ensure areas in between the teeth are cleaned.
- 4. Avoid brushing against the gingiva to avoid recession.
- Calculus is formed from long standing plaque. It can only be removed professionally.

## APPENDIX IV: INTRA-OBSERVER RELIABILITY

# Tables 33 and 34: Agreement of the 1st and 2nd readings in the

#### measurement of attachment level

BASELINE RESULTS

2nd reading													
	Q	1	2	3	4	5	6	7	8				
0	39	9	3										
1	9	63	15	1									
2	1	10	46	5	1								
3		1	16	14	3								
4			2	3	3					1s <sup>-</sup>			
5				1		1	1						
6						2							
7					2	1							
8								1					

1st reading

Agreement of the first and second reading occurred in 66% of the sites. The first and second readings differed by  $\pm 1$  mm in 28%,  $\pm 2$  mm in 4% and  $\pm 3$  mm in less than 1% of the sites.

MEDICAL LIBRARY UNIVERSITY OF NAIROBI P. O. Box 19676 NAIROBI

## FINAL RESULTS

2nd reading 0 93 13 5 15 0 11 1st reading 

Agreement of the first and second readings occurred in 75% of the sites. The first and second readings disagreed by  $\pm 1$  mm in 22% and  $\pm 2$  mm in 6% of the sites

# Tables 34 and 35 : Agreement of the 1st and 2nd readings in the measurement of pocket depth

BASELINE RESULTS

		2r	nd 1	read	ling	5					
	0	1	2	3	4	5	6	7	8		
0	2	3					N				
1	3	30	17								
2	1	7	73	14							
3		1	26	67	4						
4				3	7					1.0+	monding
5					2	4				1150	reading
6											
7							1	1			
8											

Agreement of the first and second readings occurred in 69% of the sites. The first and second readings disagreed by ±1 mm in 11% and ±2 mm in less than 1% of the sites

			2nd	1 re	eadi	ng					
	0	1	2	3	4	5	6	7	8		
0											
1	21	10	2								
2	11	62	17								
3	1	1	9	43	1	1					
4				2	1					1st	reading
5					1						
6											
 7											
8											

FINAL RESULTS

Agreement of the first and second readings occurred in 69% of the sites. The first and second readings disagreed by  $\pm 1$  mm in 28%,  $\pm 2$  mm in 2% and  $\pm 3$  mm in less than 1% of the sites

MEDICAL LIBRARY UNIVERSITY OF NAIROBI P. O. Box 19676 NAIROBI

# Tables : Agreement of the 1st and 2nd readings in the measurement of calculus

BASELINE RESULTS 2nd reading

	0	1	2	3		
0	88				-	
1		12			1st	reading
- 2	9		22	1		
3						

There was agreement in the 1st and 2nd readings in 88% of the sites. Most of the disagreement occurred in the measurement of score 0 and score 2 calculus. Nine sites were measured to have score 2 calculus in the 1st reading and score 0 in the second reading. One site was read as score 2 calculus in the first reading and score 3 in the second reading.

FINAL RESULTS

		2nd	l rea				
the state of the s		0	1	2	3		
and the second se	0	93		10			
	1		7			1st	reading
	2	11		61	1		
	3				4		

There was agreement in the 1st and 2nd readings in 88% of the sites. Most of the disagreement occurred between score 0 and score 2 calculus.